AN ECONOMIC APPRAISAL OF THE RUBBER PROCESSING INDUSTRY

IN BENDEL, OGUN AND ONDO STATES OF NIGERIA

BY

ADEBIYI ADESIDA B.Sc. Agricultural Economics And Extension (Ibadan) 1974.

A Dissertation in the Department of Agricultural Economics

SUBMITTED TO THE FACULTY OF AGRICULTURE AND FORESTRY IN PARTIAL FULFILMENT OF THE REQUIREMENTS FOR THE DEGREE OF MASTER OF PHILOSOPHY (AGRIC. ECONS) OF THE UNIVERSITY OF IBADAN.

SEPTEMBER 1977

ANLIBRARY TO MY LOVING MOTHER MAMA KUNER

ABSTRACT

This 'study attempts to assess the economic performance of the rubber processing industry in Bendel, Ogun and Ondo States of Nigeria. The Nigerian rubber industry which experienced a boom in the early 1950's started to face some serious set-backs during and after the civil war which broke out in 1967. This setbacks were manifested in form of declining productivity and consequent decline in its contribution to foreign exchange earnings. The study which covers the period 74 75 assesses the economic performance of some rubber processing firms in the three states mentioned above in order to identify some of the factors responsible for the set-backs in the industry. The data used for the study were collected mainly through the use of questionnaires and personal interviews. Other sources of data include Federal office of statistics and ministries of Agriculture and Natural Resources in all the three states. Chapter 1 assesses the economic importance of rubber industry

in terms of foreign exchange earnings, employment generation and value added to the Nigerian economy. The rubber industry contributed 6.2 percent of the country's total export earnings in 1963. By 1972, it accounted for 4,7 percent of the country's industrial labour force and Chapter 6 uses non-parametric statistical technique to assess the factors limiting expansion and economic performance of the industry. The major limiting factors are lack of capital and inadequacy of raw materials. Labour is not a limiting factor although the industry is short of skilled and managerial personnel.

A financial analysis for the industry and the estimation of a profit function for rubber processing industry were the major highlights of chapter seven. The financial analysis shows some weak points in financial management positions of some firms while the profit function shows that volume of sales, and selling expenses are the two major factors affecting profits in the processing industry.

Some policy recommendations were advanced and these include the need for individual processors to strive to make use of their under utilized capacity instead of plant expansion. Also, each processor must upgrade his management capability. The rubber commodity board should (i) appoint a committee to look into the present state of affairs of the rubber industry, establish rubber lumps marketing centres and make provisions for price incentives to producers of natural rubber at all levels.

- i'ii

ACKNOWLEDGEMENT

I am greatly indebted to Professor Olayide, my able head of Department for his much appreciated effort to see that I undergo this graduate work.

My whole-hearted thanks go to Dr. S.A. Oni, my Supervisor, for the interest he has shown in me throughout my career as a student and the pains he took all through this work. His constructive criticisms and suggestions have been very valuable.

I also wish to extend my gratitude to Dr. Akinwumi for his efforts in making this study a success and Dr. Ikpi, members of my dissertation committee for offering helpful suggestions and assistance at various stages of the study.

I have been associated with several people and organisations during this study. I wish to express my sincere gratitude to Mrs. Oni and Mrs. Osofisan both of Computer centre of University of Ibadan for helping to write-up and run the programmes for the study. I am partifularly grateful to Mr. Owumi, the Managing Director of Okpe Trading Company, Amukpe for his valuable historical background information of the Rubber Industry. My thanks also go to all rubber processors especially Mr. Quinn, the Manager of Lewis and Peat (Nig.) Limited. Ogbarafe for giving me a free access to their firms' records.

iv

I am grateful to the various Officers in the industrial Survey of the Federal Office of Statistics Lagos, the States Ministries of Agriculture and Natural Resources for providing relevant information to the study.

Next, I must express my appreciation to Professor and Mrs. Dupe Olatunbosun who have contributed immensely to my education. My appreciation goes to my beloved uncle Mr. Adebobajo Adesida for his fatherly love, who apart from being responsible for my education, bought me a car to facilitate my graduate study. My thanks also go to my brother, Mr. Adewole Adesida for his wonderful fatherly affection. I am greatly indebted to my most understanding and kind uncle, Adetiloye Adesida, for his personal interest in me.

Also, I wish to express my appreciation to Miss Rhod Okugbe of IITA, Mrs. Damilola Adeniyi of CRIN, Mr Olaleye Audu, Miss Ayomide Ige, and all the office staff of Department of Agricultural Economics, University of Ibadan for their patience and Co-operation in providing secretarial services.

Finally, I must express my appreciation to my wife, Moji and our daughter, Tutu, who gave me all the assistance during hard-time and encouragement throughout the period of this study.

V

I certify that this work was carried out by

Mr, Adebiyi Adesida in the Department of

Agricultural Economics, University of Ibadan.

SUFERVISOR S.A. Oni, B.Sc., M.Sc. (Calif), Ph.D. (Ibadan) Senior Lecturer in the Department of Agricultural Economics, University of Ibadan - Nigeria.

TABLE OF CONTENTS

			PAGE
ABSTRACT	0.6 X	***	1 - 1i
ACKNOWLEDGEMENTS	***	***	iv + v
CERTIFICATION	***		vi
TABLE OF CONTENTS		R	vii
LIST OF TABLES		ו	xii
LIST OF FIGURES		·	xiii - xvi
CHAPTER I	OF.		
INTRODUC	DTION		1
The Economic Importance of Rul	ber in Nige	rla	1
General Consideration			1
Employment Generation			5
Value added by Rubber Industry		• • •	6
Secondary Benefits	• • •		9
Historical Development of Rubbe	er Processin	g in Niger	ia 10
Need and Justification of study	•••	* * *	15
Objectives of Study	* * *	• • •	18
Plan of Thesis			19

CHAPTER II

REVIEW OF LITERAT	CURE, METHO	DOLOGY AND	DATA SOURCES	21
Review of Previous	studies	•••		21
Methodology			2	23
Area of study			··· 2	23
Selecting the samp	le			24
Sources of Data	* * *		No.	25
Primary Sources	•••			20
Secondary sources		· 20.		26
Limitations in Data		\$.`		27
The techniques of I	Analysis			28
CHAPTER III	2			
BACKGROUND INFO	ORMATION OF	THE SAMPLE	D RUBBER	27
The location of the	Rubber proce	ssing firms		29
Factors determining	g the location	of each firm	in the	
industry		* * *		20
Types of Business	Organisations	3		31
Capital Investment	Pattern			33
Employment structu	ire			3.1
The major inputs in	n rubber: proce	ssing		38

viii

FACE

	•		
CHAPTER IV			PA JE
COSTS AND RETURNS ANALY	YGIS IN RUBBER	PROCESSING	61
Analysis of Costs			61
Theoretical considerations	5 • • • #	* * *	61
thodology		🗸	63
Cost of rubber processing			64
Cost of raw material	* * 8	<u>~</u>	67
Labour cost	• • •	2.	57
Selling and Shipping expen	nses		68
Fixed cost			69
Returns analysis		•••	70
CTAPTER V	ð i		
COST FUNCTIONS FOR RU	BBER PROCESSI	ING INDUSTRY	92
Preliminary considerations			92
The Basic Model			91
Cost-output relationship			94
Data sources and Empirica	al Results		95
Optimum point of Production	on		101
CHAPTER VI			
FACTORS LIMITING THE E	XPANSION AND	ECONOMIC PER-	-
FORMANCE OF THE RUBBE	R PROCESSING	INDUSTRY	105
Factors limiting expansion	1		105

*

			FAGE
Interpretation of results			111
Factors limiting the Econom	ic performa	nce of the	113
Leele of Control			110
Lack of Capital			113
Jhortage of spare parts and	maintenanc	e workers 📿	115
Shortage of raw materials	* * *	· So.	115
Labour Problem		Sec. 1	117
Marketing Problem		V	118
CHAPTER VII		F	
AN ANALYSIS OF FINANCIAL	STATEMEN	TS AND ESTIMATION	
INDUS	TRY	11100205210	119
Freliminary Consideration	4		119
Financial Ratios and their u	ise in Busin	ess Management	110
Types of financial ratios	* * *		121
Profitability ratios			121
Rate of return on operating	Capital		122
Ratio of fixed cost to total	revenue		12.1
Rate of return on volume of	sales		120
Liquidity Ratios			125
Solvency Ratios			130
Analysis of Factors Influence	cing profits	in the rubber	
processing firms			131

4			PAGE
The basic model			136
Eypotheses			137
Estimating procedure			136
Empirical Results			139
Eliminating autocorrelated	d errors from the	model	142
Empirical results of the	adjusted model	No	145
CHAPTER VIII		2	
SUMMARY /	AND CONCLUSIO	ONS	151
Summary of Major Find	lings		151
Limitations of the study	· · ·		160
General conclusions an	d policy recomm	endations	161
References			165
Appendix I	* * *		172
Appendix II	* * *		179

2

xii

xiii

LIST OF TABLES

Table		PAGE
1.1.	Comparative earnings from Nigeria's major	2
1.2	Volume and value of rubber exports from Nigeria (1960 - 1973)	3
1.3	Percentages change in volume and value of total export of rubber from Nigeria (1960 - 1973)	4
1.4	Employment in the rubber manufacturing industry compared with total employment in all industrial establishments in Nigeria (1965 - 1972)	7
1.5	Value added for the rubber manufacturing establishments employing ten or more persons in Nigeria (1965 - 1973)	8
1,6	Processing firms in the area of study as at December, 1976	16
3.1	Location of rubber processing firms state - wise comparison	30
3.2	Factors determining the location of each firm in the industry	31
3.3 >>	Types of Business Ownership	32
3.4	Sources of Investment funds	33
3.5	Employment distribution by categories in the firms in 1974 and 1975	34
3.6	Employment distribution by categories in each firm 1974 and 1975	35

xiv

Table		PAGE
3.7	Employment distribution by categoris in each firm 1975	36
3.8	Size of firms as determined by number of employees 1974 and 1975	37
3.9	Size of firms as determined by value of Turnover 1974 and 1975	37
3.10	Size of firms as determined by the present value of assets 1975	38
3,11	Sources of Raw materials	40
3.12	Types of Finished products	42
3,13	The setting of rubber prices	58
4.1	Cost outlay for the rubber processing firms in 1974	71
4.2	Cost outlay for the rubber processing firms in 1975	72
4.3	Cost outlay for the rubber processing firms percentages 1974	73
4.30	Miscellaneous cost outlay 1974	, 74
4.4	Cost outlay for the rubber processing firms in percentages 1975	, 75
4.4a	Miscellaneous cost outlay 1975	76
4.5	Labour cost in relation to total cost in 1974	1 77
4.6	Labour cost in relation to total cost in 1975	5 78

4.7	Raw materials cost in relation total cost 1974 cost 1974	79
4.3	Raw materials cost in relation to total cost 1975	03
4.9	Selling expenses in relation to total cost 1974	81
1.10	Selling expenses in relation to total cost in 1975	82
4.11	Fixed costs in relation to the total cost in 1974	63
4.12	Fixed costs in relation to the total cost in 1975	84
4.13	Gross receipts, total cost and net revenue 1974	86
4.14	Gross receipts, total cost and net revenue in 1975	67
4.15	Gross receipts, cost and net returns for the processing firms in 1974	88
4.16	Gross receipts, cost and net returns for the processing firms in 1975	89
4.17	Net returns per tonne of processed rubber in all the processing firms in 1974 and 1975	91
5.1	Cost functions for rubber processing industry TVC/Q 1974	97
5.2	Cost functions for rubber processing industry TVC/Q 1975	9 8
5.3	Cost functions for rubber processing industry TC/Q 1974	99
5.4	Cost functions for rubber processing industry TC/Q 1975	100

xy,

Table

PACE

Table			PAGE
6.1	Ranks assigned to the factors at the expansion of the rubber proc industry in order of importance	fecting essing 	103
6.2	Analysis of ranks assigned to the statements by sixteen rubber pro	e eight cessors	2-110
7.1	Rates of return on operating cap 1974 - 1975	ital	123
7.2	Ratios of fixed cost to total reve	nue	125
7.3	Rate of returns on volume of sal	es	127
7.4	Liquidity ratios of some firms in processing industry	the rubber	12
7.5	Solvency ratios of some selecte the rubber processing industry	d firms in	132
7.3	1974 Profit functions for rubber industry	processing	130
7.7	Correlation matrix for the select affecting profits	ed factors	1/1
7.5	Frofit functions for rubber proce in 1974	ssing indus	try 1 44
7.05	Profit functions for rubber proce in 1974	ssing indus	try 1:5,
7.10	Correlation matrix for the select excluding volume of sales	ed factors	145
7.11	Profit functions for rubber proce industry in 1974	ssing	147

xvi

Table PAGE 7.12 Profit function for rubber processing industry in 1974 148 7.13 Correlation matrix for the selected factors excluding volume of sales 149

xvii

xviii

LIST OF FIGURES

MIERSI

FIGURE		4	PAGE
I	Chart showing rubber sheet proc	essing	46
II	Chart showing crepe rubber proc	essing	52
III	Chart showing crumb rubber pro	sessing	56
IV	Chart showing the marketing and	d distribution	
	channels of rubber	•••	59
	A -		
	S		

CHAPTER 1

INTRODUCTION

1. THE ECONOMIC IMPORTANCE OF RUBBER IN NIGERIA

A. General Consideration

Prior to the advent of petroleum on the Nigerian economic scene, rubber ranked fourth as Nigeria's most valueble agricultural export commodity. In 1960, rubbes accounted for about 8 percent of its total export earnings and in terms of aggregate figure, rubber export was estimated as ¥28.5 million.

Table 1.1 below shows the comparative importance of rubber earnings in relation to other major export commodities. It can be observed from this table that the percentage contributions of rubber to Nigeria's export earnings have declined drastically from about 8.4% in 1960 to 0.57% in 1974. This decline in percentage contribution of rubber is primarily due to the fact that the entire agricultural exports have been relegated to the background following the upsurge in petroleum export as well as the expanding domestic market for rubber leading to reduction in total rubber exports.

Nevertheless, rubber is still a significant agricultural commodity in Nigeria, particularly so in Bendel State where about 90 percent of Nigerian rubber output is produced.

Table 1.2 summarizes the trend in volume and value of rubber exports in Nigeria. It can be noted from the Table that the volume of rubber export

Table 1.	1. Com	parative	earnings	from	Migeria	s major	agricultur	al execrts
Window and William and Art I and Antoine and	CONTRACTOR OF A DESCRIPTION OF A DESCRIP	All summaries on the second se		and the second of the second s	Presidentian California and and	the restored is the second of the second second	Constrained in second constrained in such as a	and the second second second second second

fear	Total Exports (Ħm)	Rubber %	Palm oil %	Palm Kernel %	Groundnut	Soco2 %	Aggregate of Five as % of Total Exports*
1960	339.4	8.4	8.2	15.4	13.5	21.6	67.1
1961	347.2	6.3	7.6	11.5	18.5	19.4	63.3
1952	337	7.9	5.3	10.0	19.2	19.8	62.2
1963	379.4	6.2	4.9	11.0	19.2	17.0	58.3
1964	429.4	5.7	5.0	9.8	16.0	18.7	55.2
1965	536.6	4.1	5.1	9,9	14.1	15.9	49.1
1966	542	2.3	4.8	2.9	13.1	20.2	43.3
.967	568.2	4.0	3.9	7.9	14.4	9.3	39.5
9.68	422.2	1.5	0.02	2.4	S.O	12.1	25.02
969	635.8	1.5	0.06	1.6	5.6	8.3	17.06
.970	1885.7	0.92	0.05	1.15	2.3	7.06	11.48
971	1293.4	0.96	0.24	2.0	1.9	11.14	16.24
972	1411.9	0.49	0.02	1.1	1.3	7.16	10.07
973	2277.4	0.85	× - >	0.82	2.0	4.9	8.57
1974	5794.7	0.57	-	0.75	0.12	42.6	4.04

SOURCE: Annual abstract of statistics and digest of statistics (various issues) F.C.S. Lagos.

* These percentages were obtained by summing up the naira value of the five export crops and finding what percent the amount is of total export value.

- 2 --

Year	Total exports from Nigeria (TONS)	Value of all exports (約,000)	Percentage of all export earnings (%)
1960	57,229	28,478	8.4
1961	55,167	22,048	6.3
1962	60,591	22,712	7.9
1963	64,152	23,576	6.2
1964	73,226	24,332	5.7
1965	68,963	21,976	4.1
1966	71,398	22,948	2.3
1967	48,646	12,694	4.0
1968	52,807	12,622	1.5
1969	57,288	19,288	1.5
1970	59,286	17,568	0.92
1971	51,072	12,406	0.96
1972	41,802	7,350	0.49
1973	49,385	19,396	0.85

Table 1.2. Volume and value of rubber exports from Nigeria (1960-1973)

SOURCE: Annual abstract of statistics digest of statistics (various issues) F. O. S. Lagos.

	Total exports	Percentage	Value (¥.000)	Percentage
Year	(TONS)	change	~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~	change
1960	57,229		28,478	-
1961	55,167	-3.6	22,048	-22.57
1962	60,591	5.87	22,712	-20.25
1963	64,152	12.10	23,576	-17.21
1964	73,226	27.95	24,332	-14.56
1965	68,963	20.50	21,976	-22.83
1966	71,398	24,75	22,948	-19.42
1967	48,646	-14.99	12,694	-55.43
1968	52,807	- 8.08	12,622	-55.68
1969	57,288	0.10	19,288	-32.27
1970	59,286	3.59	17,568	-38.31
1971	51,072	-10.76	12,406	-56.44
1972	41,802	-26,96	7,350	-74.19
1973	49,385	-13.71	19,396	-31.89

Table 1.3 :	Percentage	change	in	volume	and	value	of	total	exports
	of rubb	er from	Nic	eria (1	960-	1973)			

SOURCE : Annual abstract of statistics and digest of statistics (various issues) F. C. S. Lagos. was on the increase until the civil war broke out in the country in 1966. Since the war broke out and even after the war, there has been a decline in both volume and value of rubber export. The decline in export earnings from rubber can be attributed to the decline in rubber prices and the rise in production of the synthetic rubber which has virtually become a perfect substitute for natural rubber. The presence of synthetic rubber also subjects the world market price of natural rubber to wide fluctuations. At present, like many other agricultural products, rubber prices vary from week to week at international markets.

In order to assess the trend in volume and value of rubber exports, Table 1.3 was computed, using 1960 volume and value as the base. It can be observed in this table that the volume shows initial increasing and later declining trend while the value shows a general declining trend which is a reflection of the falling prices obtained for natural rubber. B. <u>Employment generation</u>

Probably, the most significant contribution of the rubber industry to the Nigerian economy is the high employment opportunities it offers to both the rural and urban population. Anschel estimated that between 184,000 to 304,000 people representing 12 to 15 percent of the Bendel state active population are engaged in peasant rubber production⁽¹⁾. Also, in terms of industrial employment, rubber manufacturing firms have always ranked high in employment generation. The industrial survey in 1972 showed that about 8,000 people were employed by rubber processing industry. This represented about 4.7 percent of the country's total industrial labour force at that time. The industry therefore goes a long way in helping the nation to achieve self-reliance and self-sufficiency in that it provides relatively large quantities of both skilled and semiskilled labour. The industrial survey also indicated that about 3.7 percent of the total labour force in the rubber industry was either skilled or semiskilled. Table 1.4 shows only the number of people employed by all rubber manufacturing firms including the plantation factories but excludes peasant smallholders.

Also, in the rural areas, a lot of skill and experience have been acquired at different levels in the industry. This embraces the smallholders, tappers and rubber curing workers. People who work in the factory acquire, over the years, the technical skill required to handle the processing machines. Almost most mills train some processing managers on the job. C. Value added by rubber industry

Another important aspect of the industry is its contribution to the country's gross domestic product in terms of value added. For the past

- 6 -

Table 1.4. Employment in the rubber manufacturing industry compared with total employment in all industrial establishments in Nigeria (1965 - 1972).

Year	Number employed by all industrial establishments (a)	Number employed by rubber industry (b)	(b) as % of (a)
1965	95,614	6,823	7.14
1967	76,395	2,382	3.12
1968	86,721	3,362	3.88
1969	102,532	6,102	5.95
1970	129,203	8,589	6.65
19 71	145,445	6,013	4.13
1972	167,470	7,660	4.57

SOURCE : Annual abstract of statistics (various issues) F.O.S. Lagos.

Table 1.5.	Value added by the rubber manufacturing
	establishments employing ten or more
	persons in Nigeria (1965-1973)

Year	No. of estab.	Gross output (种,000) (a)	Industrial costs (钟000) (b)	Value Added (并,000) (a-b) = (c).
1965	35	22,882	12,994	9,888
1967*	21	12,618	6,456	6,162
1968*	25	17,918	9,474	8,444
1969*	31	27,022	\$4,744	12,282
1970	33	34,312	< 18,640	15,672
1971	27	59,924	35,036	24,888
1972	35	37,490	20,333	17,157
1973	29	39,755	19,423	20,332

SCURCE Annual abstract of statistics (various issues) F.O.S. Lagos.

Excludes Imo, Anambra, Cross River and River States of * Nigeria.

ten years the rubber manufacturing sector has shown some encouraging trends in its value added in the economy. This may be due to some protection and other industrial incentives accorded the firms in the industry.

Table 1.5 summarizes the value added by the rubber industry between 1965 and 1973. The term value added is here defined by the Federal Office of Statistics as the naira value difference between the value of gross output and the industrial costs of the industry. The value added by rubber industry rose from #9.8 million in 1965 to #24.9 million in 1971 but dropped to #20.3 million in 1973.

D. Secondary Benefits

The rubber industry like most other industry provides two types of benefits. These are the direct economic benefits which can be measured in naira value as discussed in section A to C and the secondary benefits which are also important for economic development purposes. The establishment of a new rubber plantation/factory is likely to be accompanied by such activities as road building, housing construction for workers; establishment of markets and the building of schools for staff children. These activities help in no small way to develop the areas and thus make life more comfortable for the people living in these areas. The rubber plantations in Bendel, Ogun and Ondo states helped to enhance the living standards of the people around the project sites and thus reduced the drift of rural people to the urban centres. An example of such plantation is the Pamol Rubber Estates in the Bendel state of Nigeria. It has well built rodds, camps for all grades of workers with both electricity and pipeborne water. Also it provides school for staff children.

II. HISTORICAL DEVELOPMENT OF RUBBER PROCESSING IN NIGERIA

The rubber processing industry is made up of a group of firms and mills that process natural rubber in the form of latex, scrap and lumps into semi-finished forms such as rubber sheets, crepe and crumbs. These are intermediate products which are either exported or consumed by local manufacturing firms. These products can be further processed into finished and final products such as tyres and tubes, shoes, mattresses, bags and many wearing apparels.

There has been a lot of questions and debate as to the exact form in which natural rubber was first exported from this country. As documented by Sara Berry, natural rubber was first exported from the country in 1885, the export being about 534 pounds and it rose over a decade to about 5,069,577 pounds in 1895. The great increase in production was due to the exploitative activities of some commercial Ghanaians who exploited and ruined all available rubber producing trees in the southern parts of the country⁽¹⁴⁾.

In an attempt to replenish the trees, government started the growing of the high yielding rubber specie <u>Funtumia elastica</u> in plantations and at the same time encouraged the villagers to establish communal plantations in suitable places near the villages. There was an enthusiastic response to this call, particularly around the Bendel state area. Many farmers took to the cultivation of natural rubber and also many private investors established large rubber plantations. Among these were JoThomas rubber estate which, according to its chairman, was established in 1899 and subsequently started to process rubber around 1925. The Millers Brothers established another rubber estate at Sapele just about the same time. Also, in 1907, an English firm established another estate at Ikot Mbo in the Cross River State of Nigeria.⁽⁵⁷⁾

Most of the rubber export from this country before the 1950s were processed by rubber smallholders. In most cases, they employed primarily very crude processing methods which in a way led to the low quality reputation earned in the international market by the Nigerian export processed rubber. Instead of using the standard coagulants such as formic acid and acetic acid, the smallholders made use of cassava starch water, citrus

- 11 -

juice from various wild trees for the coagulation of their latex. Usually, a bottle was used to roll the coagulum until the required thickness was attained. This would then be passed several times through the handmangle (smooth roller) until the sheet gave a standard thickness of about 1/8-1/4 of an inch. Smoking of the sheets was carried out by the smallholders in a kitchen combined with the drying in the sun whenever the weather was bright.

The then prevalent poor quality made the former Western Region government (which was made up of Bendel, Ogun, Oyo, Ondo and parts of Lagos states) to step up its development activities concerning the entire rubber industry. It emberked on various programmes to train peasant rubber holders, tappers and processors. Seeing that not much success was achieved, government started to establish co-operative processing factories in the villages. Government's aim was to assist the smallholders in getting better priced sheets by using modern processing techniques. This was welcomed by the farmers and many villages were formed into co-operative processing societies.

In order to encourage indigenous enterprising private investors, government decided to set up a model sheet rubber factory at Benin. This was to demonstrate to the people the gains and benefits from factory rubber processing. So between 1950 and 1960, there were many commercial rubber processing factories in the region including six rubber plantation factories. Almost all the factories were engaged in rubber sheets production but with time many either changed completely to or incorporated crepe rubber production. Although, this has lower selling price, it was easier and less complex to manufacture. In addition, crepe rubber has a relatively lower cost of production.

In the late 1960s, smallholders' manual methods of production of rubber sheets and crepe were virtually discontinued. They instead concentrated mostly on the production of rubber lumps to feed the existing mills. The practice of paying a flat price on the basis of weight without much regard to quality, coupled with the low prices of lumps encouraged the lumps producers to adulterate their rubber lumps. The presence of these adulterants which could not be removed during the processing of lumps into crepe rubber was also in part responsible for the low grades of crepe rubber.

In the last decade, especially during and after the civil war, many processing firms had folded up due to the acute shortage of raw materials.

in the industry. In 1966, Lloyd ⁽³⁶⁾ reported that about 18 crepsing factories were in operation and Ckereka ⁽³⁹⁾ in the middle of 1969 stated that there were 13 crepsing factories excluding the plantation factories. At the time of the present survey (November - December 1976), there were about 14 factories including 5 plantation factories in the Bendel State that were engaged in either rubber sheets, creps or crumbs production. Out of these, only 2 mills were established after the civil war, while in Ogun and Ondo states there were only 3 plantation integrated factories.

At the time of this survey, over 80% of the non-plantation factories are working below 50% of their installed capacity as a result of shortage of raw material. If this continues for the next decade, probably only the plantation mills would continue to process rubber, the other marginal factories having being forced to close down.

Another aspect of the problem peculiar to the industry is the diversification of products. As mentioned earlier, rubber sheets processing was the first to be introduced into the country, followed by crepe production. The changing to crepe production involved the purchase of new sets of crepeing machines and other capital necessary investment. Between 1969 and 1970, however, some firms as a result of a new type

- 14 -

of rubber product termed rubber crumbs developed in Malaya, commenced production of rubber crumbs. The technological process differs a little from those of rubber sheets and crepe, hence new capital investments had to be made by any existing firms that wished to produce crumbs. Crumb rubber has some advantages over other products. It can be put into a variety of uses. While it takes about 4-6 days to produce rubber sheets and more time to produce crepe, crumb rubber comes out in 5-3 hours. It is now being produced in the three states studied by the author:

Table 1.6 shows the year of establishment and the location of all the existing rubber processing firms in the Bendel, Ogun and Ondo states of Nigeria by December 1976. It can be observed from the table that over 80% of the existing mills are situated in Bendel State. This further stresses the important role played by the state as the major natural rubber producer in the country. These mills were established at different periods of this century. However, the bulk of the firms were established between 1950–1970. Ogun state has two mills, while Ondo state has only one factory. The three mills were established in the 1960s by the former Western State government.

III. NEED AND JUSTIFICATION OF STUDY

In the last two decades, the Nigerian Rubber Industry has been facing some problems. Considering all the government's efforts to increase

- 15 -

	and the second		
Firm code	Years of establishment	Location	State
A 01	1957	Sapele	Bendel
A 02	1971	Amukpe	19
A03	1925	Ogbarafe	S.
A04	1959	Amukpe	0
A05	1959	Benin	R
A06	1962	Ogbara	a.
A07	1960	Amukpe	11
808	1961 🏑	Amukpe	11
A09	1969	Utagba-Umo	0
A10	1955	Benin	11
A11	960	Urbonigbe	11
A12	1908	Ogbarafe	
A13	1961	Benin	112
A14	1964	Ilushin	Ogun
A15	1967	Ikenne	U.
A16	1962	Araromi	Ondo
*A17	1974	Benin	Bendel

Table 1.6: Processing firms in the area of study as at December, 1976.

SOURCE : Field survey (November - December 1976). *Started production in June 1975. Hence not included in the study. rubber yield, its foreign exchange earnings over the years showed a declining trend. This in part has been attributed to the world market prices of natural rubber which keep on fluctuating from week to week.

Another major problem which stems from the rubber industry itself is the shortage, in recent times, of raw materials such as latex, lumps, and scraps. All these problems have important implications for the firms in the rubber processing industry.

Many firms had folded up due to the lack of regular supply of natural raw rubber. The problem was aggravated at the outbreak of the civil war when most of the itinerant rubber tappers of Bendel state had to leave for the Eastern states. Between 1966 and 1968, out of the 18 factories in Bendel state, 5 were forced to close down due to the lack of raw materials. Even after the civil war was over, the rate at which processing firms are being established is far less than the rate of exit. As for now, virtually all the smallholder-processors and co-operative processing societies have folded up.

The rubber processing industry in a way links the natural raw rubber producers with the consumers of rubber sheets, crepe and crumbs. These can either be manufactured locally into finished products or exported. Considering its important role in the whole rubber economy, in order to forstall the present decline in the rubber industry, there is an urgent need to

- 17 -

re-appraise the existing rubber processing firms in light of their operational performance. Such appraisal will provide the necessary framework for guidelines on future policy on the rubber industry in Nigeria.

IV OBJECTIVES OF STUDY

The major objective of this study is to appraise the economic performance of all the rubber processing firms in Bendel, Ogun and Ondo states of Nigeria. This broad objective will be fulfilled by examining the following specific objectives:

- To examine the pattern of capital investment among the rubber processing firms.
- To assess the employment structure within the rubber processing industry in terms of skilled and unskilled labour, Nigerians and non-Nigerians.
- To determine the costs and returns involved in rubber processing business of the three states.
- 4. To derive a cost function for Rubber processing industry and also identify the factors militating against the expansion of the industry.
- To carry out a financial analysis of the industry and estimate the determinants of profit within the rubber processing industry.
Finally, to make appropriate policy recommendations towards improving the economic performance and expansion of the industry.

PLAN OF THESIS

In the rest of the thesis, chapter two deals with the review of previous studies and the methodology adopted in the study. The review examines all available, relevant studies, reports and past work dealing with rubber processing. The Research Methodology describes the area and scope of field investigation, sampling procedures, data sources and the limitations of data used.

Chapter three presents some background information of the rubber processing firms. This information includes the types of ownership, the location, sources and pattern of capital investments, employment structure and technology involved in rubber processing.

This includes some theoretical considerations as well as empirical input-output relationships giving details of cost profile, gross and net returns in the industry.

Chapter five analyses the cost function of the rubber industry, while chapter six discusses some factors limiting the expansion and

Er-

Chapter seven is concerned with a financial analysis of the rubber processing industry. This chapter is involved with the derivation of some financial ratios as well as empirical estimation of the determinants of profit in the rubber industry.

The last chapter presents a summary and conclusion of the study. This conclusion also highlights some policy recommendations for the expansion and better economic performance of the industry.

CHAPTER II

REVIEW OF LITERATURE, METHODOLOGY AND DATA SOURCES

This chapter presents a review of previous studies, reports and past works dealing with the Nigerian rubber Industry with particular reference to rubber processing. It also outlines the methodology adopted and the data sources.

A. REVIEW OF PREVIOUS STUDIES

A lot of research studies have been carried out in the past on the Higerian Rubber Industry. Almost all these studies focussed attention on the economics of rubber cultivation. The rubber processing sector, which is an integral part of the whole industry has however been neglected or unexplored from the view point of research. This is why it was not possible to find an earlier study that dealt specifically with the rubber processing industry in Nigeria. Some of these studies mainly touched on the pattern of production and government developement programmes while some dealt with marketing.

The study by Anschel⁽⁸⁾, gave a general background to the rubber industry with particular reference to the economics of peasant rubber production in Nigeria. It is observed from his study that the smallholders cultivated about 95% of the rubber farms and produced 85% of total rubber output in the country. Secondly, he confirmed that most of the smallholdings are made up mainly of unselected seedlings and wild rubber species which yield relatively low quantity of latex. Crude tapping methods also contribute to lower yield of rubber. Despite all these, Anschel concluded that small holdings are more profitable than large rubber estates due to the large overhead expenses on the rubber estates.

With regards to rubber processing, Anschel identified some rubber processing factories, but he did not analyse the economics of factory rubber - processing. He only discussed smallholders manual rubber processing.

A more relevant study about rubber processing is the work of Patrick Gromely⁽²⁵⁾. He gave a descriptive account of the economic behaviour of the rubber processing sector of the industry. He regarded the processors (at the time of his study) as being Oligopsonistic since he believed that they had power to influence the price of rubber lumps and scraps which they processed. This he attributed to the fewer number of processors unlike the very numerous smallholders that produced mostly adulterated rubber lumps. He in effect advocated the expansion of the rubber processing industry rather than cut down the number of the existing mills.* He also suggested that public authorities such as the newly

- 22 -

^{*} The reduction in the number of the existing processing units was strongly recommended in 1968 by a committee headed by the Obe of Benin appointed by the Mid-Western State Government.

established Bendel State Rubber Development Agency (BRDA) could enter the processing industry and compete with the existing processing firms.

Another relevant study is that of Obi⁽³⁸⁾. This study concentrated on the government owned crepe factory at Ikpoba, Benin City. Although, the study sportlights some of the broad problems of the whole rubber industry with particular focus of attention on the problems facing the processing factories. The findings and recommendations are two generalized to be meaningful and useful for the rubber processing industry.

Other available studies include the work of Okereka⁽³⁹⁾, Laurent⁽³³⁾, Saylor⁽⁵²⁾, Sawyer⁽⁵¹⁾ and Zwanhunsen⁽⁶²⁾. These studies were rather general in scope and none of them focus attention on the economic appraisal of existing rubber processing firms in the country. This is why the present study is designed primarily to appraise the performance of the existing rubber processing firms in Nigeria.

would increase the demand for raw rubber and so encourage increasing production of rubber.

B. METHODOLOGY

Area of Study

The study was carried out in Bendel, Ogun and Ondo States. These states were chosen primarily for three reasons. Firstly, Bendel State

is the major rubber producing state in the country and should therefore have the greatest number of rubber processing firms.

Secondly, the three states (i.e. Bendel, Ogun and Ondo) including Oyo state and part of Lagos State comprised the former Western Region. In that wise, they would all have many things in common. Many policies made concerning the rubber industry were made by the old Western Region of Nigeria. Such policies included the establishment of many of the existing large rubber plantations, co-operative processing units and the rehabilitation of the existing rubber holdings and various other development. programmes.

Lastly, apart from financial and time limitations, the three states account for over 94% of the output of rubber in Nigeria while the other Eastern States where rubber is also produced (comprising of Cross River, Anambra, Rivers and Imo States) account for only 6% of the country's rubber output⁽⁴³⁾

Selecting the sample

The list of all the rubber processing establishments in Bendel, Ogun and Ondo States was obtained from the industrial survey section of the Federal Office of Statistics, Lagos. A more current and compreby the Tree Crops division of the Bendel State Ministry of Agricult te and Natural Resources. At the time of the writer's field survey there were only fourteen operating processing establishments in Bendel State. In Ogun state, there were only two processing units while Ondo state had only one establishment.

In the three states, all known existing processing units were studied. Two categories of processing units were identified. There were the processing units which are integral parts of large rubber estates and which obtain their raw materials from the estates. The other category include those mills which do not belong to any rubber estate and which obtain their raw materials mainly from local small holders.

All the identified firms in the two categories studied were found to fulfil some industrial survey specifications stipulated by the Federal Office of Statistics; Firstly, the size of the establishment with respect to the minimum capital investment must not be less than ¥10,000. Also, the level of employment must not be less than 10 employees. Thus, in this study, no smallholders or local manual-processor wes considered.

C. SOURCES OF DATA

The data used for this study were obtained from two main sources, namely primary and secondary sources.

- 25 -

Primary Sources

The records kept by individual firms served as the major source of the data. This was collected during the writer's field investigation with the aid of questionnaires. The various firms were visited in person together with two field enumerators during the months of November and December, 1976. Each firm gave appointment dates to the author for personal interviews. This was at the convenience of the senior officer granting the interview. During the interviews, the respondents were given enough time to discuss fully their processing business, and other factors influencing their economic performance. Also, at such interviews, the investigator was able to obtain some brochures, pamphlets and other relevant documents, all of which enhanced our background knowledge of the processing firms under study.

Secondary sources

Some basic data were also obtained from publications by the Federal Office of Statistics and the Federal Ministry of Industry, Lagos. The Bendel State Ministries of Agriculture and Natural Resources, Finance and Economic Development also provided useful information. Other institutions visited were the defunct Western State's Agricultural Investment Corporation and the Ogun and Ondo States Ministries of Agriculture and

- 26 -

Natural Resources. Very useful verbal information were also collected from senior employees of the above institutions. Finally, those processors who have been in the rubber industry for a very long time provided some useful historical information about the entire rubber processing industry in the country.

Limitations of data

Despite all the skills and caution of the investigator in soliciting and enlisting the support and co-operation of the respondents, some vital information was still missing. Some of the processors evaded questions pertaining to the firms' profits, tax rates, initial capital investment and the present total paid up capital of their establishments. The respondents always became uncooperative when the interviewer tried to persuade them to supply these data.

In view of the above mentioned problems, the investigator had to use some indirect procedures to obtain information on profits and all the operating expenses. The use of such indirect procedures necessarily limit the reliability of our data and the extent to which these data can be used for broad policy recommendations. However, when these data were cross-checked with those supplied by the same firms to the Federal Cffice of Statistics, no major variations were found for each firm for the period covered in this survey. WERSTH

The techniques of analysis:

In the analysis, the author employed three statistical techniques, budgetary namely: tabular, non-parametric and multiple regression. The/technique was used for the economic analysis in chapter four while details of the non-parametric and regression techniques are presented in chapters five, six and seven. Also, in chapter seven, the author computed some financial ratios for the firms which gave the needed data.

CHAPTER III

BACKGROUND INFORMATION OF THE SAMPLED RUBBER PROCESSING FIRMS

A. The location of the rubber processing firms

All the rubber processing firms included in the study are pattern located in the southern part of Migeria. This location can be exclained by two major factors. Firstly, high rainfall is a necessity for high latex yields and this has led to the restriction of rubber production to the southern rain belt. Secondly, latex, the major input of the industry is very bulky. As a result of this bulky nature, most processing firms rationally locate their plants near the sources of latex supply.

The location of the selected processing firms has been analysed and presented in table 3.1. It can be observed from this table that about 82 per cent of all the firms are located in Bendel State whilst about 12 per cent are located in Ogun state with only 6 per cent located in Ondo State.

Factors determining the location of firms in the industry.

As regards the factors that determine the choice of particular site for each mill, seven factors were specified as major determinants. The top executive of each firm was asked to identify these factors and the results of their responses are summarised in table 3.2. It can be

Table 3.1. Location of rubber processing firms state-wise compo-	ris	00
--	-----	----

State	Number	Percentage
Bendel	14	82.35
Ogun	2	11.76
Ondo	SP	5.89
Total	17	100

Source : Field Survey (November - December, 1976).

observed from the table that all the sixteen responding firms indicated "closeness to the source of raw materials" as the major determining factor for the location of the factory. Next to this was "access to roads, rail and seaports". The latter factor is important in the transportation of raw materials and the evacuation of finished products.

Other important factors are availability of water and electricity which are essential for the continuous operation of the rubber processing mills.

Factors	No. of Firms	Percentage
Cheap labour	9	56.25
Close to raw materials	16	100
Access to roads, rails		N.
and seaports	14	87.5
Where land is available	3	18.75
Availability of water	Sr	
and electricity	12	75
Government recommen-		47.75
Industrial estate	7	47,75
Others	-	-

Table	3.2.	Factors	determining	the	location	of	each	firm
			in the ir	idus	try			

Source Field survey (November - December), 1976.

B. Types of business organizations

The rubber processing industry like all agro-allied industrial establishments con broadly be classified into five major categories of ownership viz:- single proprietorship, patnership, private limited liability company, wholly government owned and quasi-government firms corporation. The ownership pattern of all the studied is presented in table 3.3:

Table 3.3. Types of Business Ownership

Type	Frequency	Percentage
Single proprietorship	2.	12.5
Patnership	4	25
Private liability company	5	31.25
Quasi - government Corporation	5	31.25
Government	X	-
Others	O.	-

Source Field Survey November - December) 1976.

It can be observed from table 3.3 that most of the firms are either private limited liability companies or quasi - government corporations. In early 1950's and 1960's, the industry was dominated by single proprietors who were most Greek nationals. As a result of the continued lack of raw materials and the Migerian Indigenization decree, many of such firms folded --up while others were sold out to Migerian investors.



- 33 -

C. Capital Investment pattern

Capital has always been a major limiting factor to many commercial ventures in this country. Even in the present days of oil - rich Minerie, capital is still not easy to come by. The Federal government in its endeavour to ease this problem has established in recent times many credit corporations. These include the Nigerian A ricultural Bank, the Ni erian Industrial Development Bank and the Nigerian Bank for Commerce and Industries. These credit corporations have made little or no contributions to the rubber processing industry as indicated in table 3.4.

Sources	Frequency	Percentage
Government	6	37 5
Personal Savings	7	43 75
Co-operative Societies	-	
Foreign Investors	6	37.5
Commercial Banks	9	56.25
Credit Corporation (NAB, NIBD, NBCI)	-	-
Cthers	-	-

Table 3.4: Sources of investment funds

Source Field survey (November - December) 1976.

The major sources of capital investment in the industry in order of importance are commercial banks, personal sources, government (these are povernment - owned firms) and foreign funds.

D. Employment structure

The employment pattern in the rubber processing firms in 1974 and 1975 is presented in tables 3.5 and 3.6. The rubber processing industry like most modern manufacturing industries is relatively less labour intensive. Almost all the operations are carried out by power driven automatio machines. These machines are however operated by semi-skilled and skilled labour. In 1974, all the firm's under study employed about 2724 oeople and in 1975, the employment decreased to 2287 persons. The detailed breakdown of the employment figures are presented in tables 3.5 and 3.6.

Table 3.5: Employment distribution by categories in the firms in 1974 and 1975

2	19	74	1975		
Category	Number Employed	Percentage	Number Employed	Percentage	
Managerial	60	2.2	53	2.3	
Clerical	162	5.9	148	6.5	
Technical	349	12.8	329	14.4	
Operatives	2153	79.1	1757	76.8	
Total	2724	100	2287	1,00	

Source Field work (November - December) 1976

1		Mana	gerial		Cleri	cal	Tech	nical	Cpera	tive	Tota	1
	Nij 1974	erian 1975	Non-N 1974	ligerian 1975	1974	1975	1974	1975	1974	1975	974	1375
	3	3	1	1	5	5	55	51	136	120	200	180
	3	3	-	-	7	7	11	11	103	101	1.24	122
- 10	3	3	-	-	6	6	25	21	56	43	90	73
- 32	1	1	3	3	2	2	10	12	120	108	136	126
. 23	1	1	-	-	3	3	3	3	68	33	75	40
.35	8	8	1	1	39	36	44	42	130	104	222	1.)1
07	11	8	1	***	1	1	3	2	188	131	204	142
÷ 38	4	3	-	-	10	3	6	6	230	60	252	72
-03	2	2	1	1	6	8	5	6	99	104	113	121
10	1	-	-		17	15	7	5	137	66	162	85
11	1	1			19	11	35	29	75	76	130	117
12	1	1	1	1	8	8	32	31	383	394	425	435
-13	2	2	1	1	1	1	10	10	47	34	61	48
~ 음식	4	3	-	-~	9	9	30	28	125	132	1 68	172
	1	1		a'	4	8	10	11	49	46	64	66
-15	5	5	- <	2	25	25	61	61	207	205	298	296
otal	51	45	Se	8	1 62	148	349	329	2153	1757	2724	2287
l of lotel	1.9	2.0	0.3	0.3	5.9	6.5	12.8	14.4	79,1	76.8	100	100
												-

3 : Employment distribution by categories in each firm 1974 and 1975

Source ; Field Survey (November - December) 1976.

- 35 -

On the whole, over 75 per cent of the workers can be classified as operatives (unskilled labour), while about 20 per cent are semiskilled and only 2.5 per cent are highly skilled. The Nigerian professionale in the industry over-shadow their expatriate counterparts in both years. The situation was attributed in part to the indigenization decree which led to the departure of some expatriate managers.

The drop in the number of unskilled labour employed in 1975 was due to the Udoji award which resulted in rise in wages and consequent decline in the number of operatives. Many casual workers were laid off whilst some were forced to seek more renumerative employment outside rubber processing industry.

Many criteria were employed in the analysis of firm sizes. These criteria include number of employees, value of annual turn-over and present capital outlay of each firm. Tables 3.7, 3.8 and 3.9 summarised our findings.

Number of	1	974	1975		
Employees	Frequency	%	Frequency	%	
Under 100	4	25	6	37.5	
100 - 199	6	37.50	8	50	
200 - 299	5	31.25	1	6.35	
300 - 399	1	6.25	1	6.25	
400 - 499	-	. Alan	-	-	
Cver 500	-		-	-	
Total	16	100	16	100	

Table 37: Size of firms as determined by number of employees 1974 and 1975

- 36 -

It can be observed in table 3.7 that none of the processing firms employed more than 400 employees in 1974 while the figure range dropped to less than 200 employees in 1975. This further confirms the fact that many people had to leave the rubber industry for more remunerative jobs. Table 3.8 : <u>Size of firms as determined by value of turnover</u> 1974 and 1975

		.0		
	1974	1975		
Annual Turnover (秋)	Frequency	%	Frequency	76
Under ₩100,000	0-A	-	-	-
₩100,001 - ₩500,000	3	18.75	5	31.25
₩500,001 - ₩1,000,000	6	37.25	6	37.5
Over №1,000,000	7	43.75	5	31.25
Total	16	100	16	100

Source - Field Survey (November - December) 1976

Also in table 3.8, the value of the annual turnover ranged between #0.1 million to over one million naira during the two years. The drop in the value of the annual turnover in 1975 was attributed by most processors to the shortage of raw materials and as a result of the various industrial actions when production in many mills had to stop.

Value of assets	1975			
(Y #)	Frequency	3 %		
Under ₩50,000	1	6.67		
₩50,001 - ₩100,000	2	13.33		
₩100,001 - ₩500,000	10	66.67		
¥500,001 - ¥1,000,000	2	13.33		
Over ₩1,000,000	-	-		
Total	15	100		

Table 3.9 : <u>Size of firms as determined by the present value</u> of assets 1975

Source : Field survey (November - December) 1976.

It can be observed from table 3.9 that the bulk of the value of the assets of the processing firm in 1975 ranged between №100,000 to №500,000. E. The Major Inputs in rubber processing

The major raw material in rubber processing industry is natural rubber. This can be obtained in forms of latex lumps and scraps. Processing firms that engage in the production of rubber sheets make use of latex while crepe rubber processors make use of rubber lumps or scraps. Latex and lumps can be used in the production of rubber crumbs. Apart from raw rubber, some chemicals such as formic acid and acetic acid are also used as coagulants in the production of rubber sheets and crumbs. Other inputs include water which is supplied by the state covernments and electricity which is mainly supplied by National Electricity Power Authority. These two inputs are of great importance to all mills.

As precautionary measures against the irrecular supply of water and electricity, most of the firms have installed stand-by generators and water tanks. Firms that process crepe use a lot of water in order to remove dirts and particles from heavily adulterated rubber lumps. In smoking of rubber sheets, some firms employ charcoal or fire woods. Fuel and all lubricating oils for the generators and other heavy machinery are other operating input which are obtainable from local oil and gas marketing companies.

F. Sources of raw materials

The two major sources of natural rubber are peasant farmers and large rubber plantations. Factories which are integral parts of large rubber estates readily obtain latex and lumps from their parent rubber estates while others obtain rubber lumps from peasant rubber holders. Another important source is through the local contractors. Table 3.10 shows the main sources of raw materials as stated by the processors.

- 39 -

Eased on personal observations during the field survey, only mills situated in rubber estates engaged in rubber sheets processing. This is attributed to the regular supply of high quality latex from the fields.

The local contractors play very important role in the industry with regards to the procurement of raw materials, particularly for mills that have no rubber plantations.

Sources	No of firms	%
Own estates	8	50
Local farmers	10	62.5
Local contractors	8	50
Other estates	1	6.25
Others		-

Table 3.10 : Sources of raw materials

Source Field survey (November - December) 1976.

The rubber contractors go to the interiors to assemble rubber lumps, and scraps of different qualities from small holders. They in turn sell to processors in lorry loads. Since there is generally an acute shortage of rubber lumps, they sell to processors who offer good prices. They also make sure that the processors are ready to pay on the spot. The old practice whereby contractors signed contracts with certain processors has virtually stopped.

G. The technology of rubber processing and products of the Industry

The industry has three major products. These are rubber sheets, crepe and crumbs. Sheet rubber is divided into standardized grades; No. IX RSS, No. 1 RSS, RSS II, RSS III, RSS IV, RSS V, RSS VI and cuttings. The grades are made from the same coagulated wet rubber sheets. Skim rubber made of skim latex is never used in whole or in part in the production of any grade.

Crepe rubber is generally regarded as lower grade of rubber. Unlike rubber sheets, it is valued according to the appearance. Light brown crepe earns better than the dark brown crepe. Crepe is made from lumps and scraps of all types and grades.

Rubber crumbs can be made from latex and lumps. Crumbs made from sheets generally have higher grades than those made from lumps and hence earn more revenue. Table 3.11 presents the types of finished products and the number of firms that process them.

Туре	No. of Firms	%
Sheet	7	43.75
Crepe	11	68.75
Crumbs	5	31.25
Field ou	1976	2

Table 3.11 : Types of finished products

Processing of rubber sheets*

Processing of rubber into sheets can be analysed under small scale or factory technology The small scale processing is employed by smallholders who employ crude processing techniques whilst factory processing employ improved techniques. The technology for processing rubber will focus attention on factory processing only, and the stages of processing include latex dilution, coagulation, milling, drying and smoking. Inspection, grading and packing are also corried out as integral parts of the processing.

A comprehensive literature on rubber processing is given by (i) Loren, J. Polhamus; <u>Rubber botany cultivation and utilization</u> pp. 179 - 208.

⁽ii) Zwanhuiwen, Th., Rural Industries for agricultural raw a aterials in Eastern Region. F.A.D. Report 1963.

Dilution of the latex

The transportation of latex from the fields to the factory sites is done in tanks. Latex is then diluted with water to a certain standard Dry Rubber Content (DRC). The dilution is required to make the rubber as uniform as possible and to obtain coagulum with a proper degree of hardness. Insufficiently diluted latex makes hard coagulum which is difficult to mill and requires excessive power. Too much dilution on the other hand makes soft coagulum which is easily torn during milling. The dilution also facilitates the settling of dirt particles after sieving and the bubbling away of gases which can reduce the sheet quality.

Coagulation in Tanks

The coagulating tanks are completely different from the transportation tanks. They are made of aluminium and hence known as all metal coagulation tanks. They usually have standardized interior dimension of 30000 x 700 x 400 mm with 75 aluminium partitions and a capacity of 76 sheets of approximately 1.5 kg dry weight when filled to a height of 34cm with latex of 16 per cent DRC. The aluminium partitions fit into notches in the side bars at the top of the tank and are kept in position by a partition spacer in the bottom of the tank.

The tanks are either fixed or movable. They are filled by means

of a latex guide between the latex receiving troughs running under the outlet cocks of the receiving tanks. After filling to the specified height, froth is skimmed off and coagulant in form of formic acid is added under constant stirring with an aluminium latex stirring rake. After the addition of the coagulant and skimming, the partitions are thoroughly wetted to prevent the air bubbles from sticking to the aluminium which cause pits on the surface of the coagulum. With care, the partitions are placed in the notches to prevent having coagulum of uneven thickness.

When coagulation has set in clear water is added to the tanks. This promotes even contraction of coaculum and prevents them sticking to the partitions and also it prevents the formation of spots on the coaculum. On complete coagulation, the tanks are pushed to the other end of the track and one by one placed directly in front of the battery. The coeculum is fed over a short chute straight into the machine between the first two rollers. This can be performed by a man.

Milling of sheet rubber and milling machinery

The sheeting batteries are a combination of four to five pairs of smooth rollers, one pair of printing rollers spirally grooved which are mechanically driven. Usually, the first pair is made up of two, 12 or 18 sided rollers to improve the grip on the thick coagulum. The gap between the rollers decreases from one pair to the next and they are geared together to run at differential speeds to take care of the alon ration as the coagulum passes through the milling battery. The rollers are driven either by an electric motor or by a Diesel engine

After passing through the final pair of rollers, the sheet is washed in a special tank at the end of the sheeting battery in which the water is constantly changed. This is to further remove the remaining serum from the surface of sheets in order to prevent the formation of function and air bubbles caused by fermentation. Where available, sheets are smoked on trucks mounted on rail tracks and these are left to drip for about an hour in a suitable cool atmosphere before pushing them into the smoke-rooms. Dripping should not be too long to prevent rustiness and internal mouldy spots.

Drying and smoking of the sheets

The sheets after dripping for about an hour still contain some 30% of moisture of which about a third is contained in the sheet and twothirds is surface water. They also contain serum with non-rubber constituents on which mould can easily develop. The purpose of curing is to preserve and to dry the sheets. Preservation is brought about by certain constituents of the smoke particularly phenols which are

- 45 -





easily absorbed by the wet sheets. Smoking also promotes a uniform colouring of the sheets and lightly coloured spots and streaks which are apparent in the smoked sheets disappear.

Asborption of smoke constituents is most rapid when the sheats are still wet. The curing process can be divided into two stages; (i) the actual smoking stage during which much smoke and moderate heat is supplied and (ii) the drying stage when increased heat and little or no smoke is supplied. In general sheet with a thickness of 3 - 3.5 cm can be cured in 4 - 6 days.

An important condition is that sheets to be smoked and dried in one batch in a smoke house should have the same thickness. A typical curing scheme for rubber sheet of about 3 - 3.5 cm thickness is given below.

First day

Smoke room temperature

Smoke production

Ventillation

- 40° for the first 3 hours rising to 45° .

- should be at full capacity.

 Bottom window fully opened and later reduced to half open while chimney should be fully opened. Second day

Temperature	-	50°C	to	55°C	
-------------	---	------	----	------	--

Smoke production - about 1/4 full capacity.

Ventillation - Bottom and chimney half open

60⁰

Third Day

Temperature -

Smoke production -

Ventillation -

about 1/4 full capacity Bottom very nearly closed

55°C - 60°C

while chimney 1/4 opened.

Fourth day

Temperature

Smoke production

- 📜 very little and slightly opened.

Strict temperature control in smoke house is very important. The temperature is recorded at regular intervals. Ideally, a sheet is considered completely dry when the rubber has become transluscent without regish opaque spots and eyes.

Smoke houses

Smoke houses are built in several ways but they all have to meet a few principal requirements. The apartments are designed to hold the total daily production. This will allow for proper smoking and drying of freshly milled rubber sheets. Smoke is produced in many ways but each smoke

- 48 -

house apartment must be provided with maximum and minimum thermometers. The walls of the smoke houses must be properly made so that as little heat as possible escapes. The production of smoke and heat and the ventillators should be simple to regulate. The compartments should be easily accessible so that charging and discharging of sheets during the smoking period if required will take the minimum of time. The roof is provided with one or more chimneys with regulating valves. The side walls of the lower part having ventillation holes should have sliding valves.

Inspection and grading of sheets

Sheet rubber grading is also carried out using appearance criterion, but unlike crepe rubber, sheets are inspected by holding them up against the light or putting them on a special table with a milk glass. The brightly lit glass facilitates the inspection for dirts, particles, air bubbles which are out away with a pair of scissors. The opaque parts which are not properly dried at the edges are cut off and they are regarded and sold as cuttings. The sheets are then sorted out into standardized grades. Packing and bailing

The rubber industry uses two sets of grades for natural rubber. Grade distinctions are made on the basis of cleanliness, quality of curing, mould and rust content. The Nigerian rubber industry adopts the malayan rubber research institute specified grades. These grades have been stated earlier. Sheets of the same grades are packed together in balas. The grades are clearly written on the bales before they are taken to the stores where they are held until they are shipped or sold locally.

Processing of crepe rubber

Crepe rubber is. made from rubber lumps and scraps. The rubber lumps are sometimes heavily adulterated. On getting to the factories, lumps are cut into two to expose the internal part and see if adulterated. Unlike few years back, processors now pay remunerative prices for lumps of good qualities. After collection the lumps are put in concrete walled tanks where they are continually soaked in water until they are ready to be milled. It is advantageous to soak the lumps in water since this softens the rubber and makes it amenable to machining.

Milling and crepeing machines

The crepeing machines are principally made up of; (i) a set of granulators, with rollers of about 12 inch diameter by 18 inch face. (ii) The macerator and the intermediate mill which are diamond grooved. (iii) A set of smoothers (finishers) with smooth rollers and (iv) a sorap washer. The scrap washer as seen in some factories consists of a heavy trough in which revolve two large and heavy rollers. These rollers have strong raised and blunt corrugations which grip the rubber and kneed it as it passes through them. The rubber is carried up again by the corrugations so that the action is automatic. A lot of water is used by the scrap. This further softens the rubber and allows the dirt to settle. The scrap washer eventually produces a disjointed but continuous homogenous wet rubber.

The cleansed rubber is taken to the crepeing machines after some 15 to 30 minutes. The wet rubber is now fed into the granulator where the rubber is turned into a continuous blanket. Water is at all times passed over the crepe blanket.

The processing is further continued on the macerator. The crepe blanket is passed through the macerator repeatedly 4 - 6 times after which it is taken to the smoothers where a crepe blanket of required thickness is produced. For the flow chart on crepe processing, see chart II. Drying of crepe rubber

Crepe rubber is never smoked. It is air dried. It has less nonrubber constituents than sheet rubber and hence it is less susceptible to mould and smoke preservation is not required. It should however be





thoroughly dried and stored in a dry place. Drying is carried out in special drying houses either by ventillation or a combination of both ventillation and heat. If no heat is applied, the drying takes about 4 weeks. A normal drying period in a heated drying house is about 8 - 12 days. The temperature measured between the crepe in a created drying house must not exceed $36^{\circ C}$ ($97^{\circ F}$).

Drying houses

A normal drying house consists of a large space divided into two or more drying floors, composed of hardwood laths with round tops. The usual vertical distance between the drying floors is 4 metres with about 50cm space between the lower drying floors. Heating of the drying houses is brought about by a number of radiators heated by steam. The radiators are mounted under a perforated floor at the bottom of the drying house. The drying houses are built in such a way as to have ventillators in the walls through where ventilation air enters the heating spaces.

Sorting Packing and Baling

Light brown crepe are sorted out and packed together. The dark brown ones which are of lower grades are also packed separately. These are baled and markings are put on the outside of each bale to differentiate the grades.

Processing of Rubber Crumbs

Crumb rubber can be made from wet rubber sheets and wet crepe blanket. The wet crepe and sheets are milled as described earlier. These are taken to the chipping machine which chips the sheets and crepe into crumbs. The wet crumbs are later put in aluminium baking pans and subsequently placed in the unifier

The Unidrier

The unidrier consists of a large oven and a compartment lying along the length of the oven. Heat is supplied to the oven from the compartment. The oven is provided with a pulley device with chain wheels on which the baking pans are placed. This allows for a smooth movement from one end of the oven to the other. The movement of the baking pans over the pulley device is automated. Heat supplied to the oven is regulated through the whole length. The effective drying of the crumbs takes about 4 - 6 hours. The crumbs come out in different grades with wet rubber sheets coming out in higher grades.

Weighing and Packing

The dry crumbs are later weighed into specified standard weights before they are taken to the pressing machines. The pressing machines compress them into very compact sizes. They are then packed in
cellophane bags after which they are put in big crates ready to be shipped. See Chart III.

E. Marketing and distribution of products

Rubber is one of the principal export crops over which the Nigerian Froduce Marketing Board has no statutory marketing control and responsibilities. Since 1950, the government has been trying to promote cooperative marketing organization among the small-holders in order to enable them secure for themselves a larger share of the price paid by the final consumer.

Recently, the government decided to create through decrees, Commodity Boards for all the major export crops of this country. This means that a Rubber Commodity Board will for the first time come into being. The board will in essence perform all the marketing functions connected with natural rubber.

Rubber sheets, crepe and crumbs are either consumed locally or exported. Over 90% of the country's annual rubber output is exported, whereas only 5 - 10 per cent is consumed locally. It has been pointed out by L.K. Laurent in his study that, prior to the Nigerian civil - war, the country exported about 72,000 tons of natural rubber, whilst only 7 percent of this were consumed locally. However, after the civil war, as a result of newly established rubber manufacturing firms, the local consumption of rubber can be expected to have increased.





- 56 -

A full account of the marketing and distribution of rubber was given by Okereka, although lots of changes have taken place within the industry after his work. Natural rubber unlike in the pre-civil war era is no bore dependent upon by many for their livelihood. The finished products are distributed by exporters and processor-exporters. Almost all the processors have direct contact with the world market. Some have marketing eigents based in London. These agents feed them with the necessary information concerning rubber international market situations. They also perform on their behalf all the marketing functions such as contracting and shipping arrangements. Rubber processors who do not have agents abroad sell through local agents.

Table 3.12 summarizes our findings on the factors determining rubber prices. Prices are set by the world market forces and the quotations are sent down by the overseas agents to local processors. In certain instances, prices may be determined by an agreement between the exporters and the processors.

Local markets

The products of the rubber processing industry are intermediate products which **are used** by some agro-allied firms for further manufacture into final goods such as tyres and tubes, bags, car mats, and other

Table 3.12 : The setting of rubber prices



Source : Pield work 1976

wearing apparels. The firms get their supplies directly from the processors and local exporters. A few large firms like Bata, Mitchelin and Dunlop have their own rubber plantations. Figure IV shows the marketing and distribution channels of the processed products.

The processors identified rising cost of marketing in relation to returns as the main marketing problem. Lack of capital to finance the marketing functions was also mentioned by the respondents. None complained about the lack of demand for their products whilst many





processors even claimed that they could not meet the demand for their products due to the lack of raw materials. This further confirmation perat the fact that many of the processors are operating below their installed capacity.

CHAPTER IV

COST AND RETURNS ANALYSIS IN RUBBER PROCESSING

The objective of this chapter is to analyse the costs and returns involved in rubber processing business. Such an analysis is quite useful in determining the present level of profits in the rubber processing industry. The unfolding structure of costs should make it possible to advise plant managers on the best way to minimize operating costs.

A. ANALYSIS OF COSTS

Theoretical Considerations

The budgetary technique has been adopted in economic analyses of crops, livestock and other agribusiness enterprises in both developed and developing countries. Bradford and Johnson ⁽¹⁵⁾ defined a budget as a written plan for future actions. A budgeting process for a given period converges on two figures; one of these is total revenue for the period, the other figure is total expenses for the same period. The difference between these two figures is called net profits or net returns if revenue exceeds costs. All other parts of budgets are regarded as details leading up to the derivation of net returns.

The principles underlying budgeting are essentially the same whether it is made for a small farm, a large plantation, an industrial firm or a government venture. An industrial firm's budget can then be defined as a plan about different kinds of financial and technical structures of a processing firm to determine its relative profitability for a given period of time. The types of technologies to budget for in a mill will be determined on the basis of available financial resources, capital assets, types of products and management capability.

In budgeting, all expected costs and returns are laid out and the quantities of input and final outputs are estimated. Then comes execution in which management uses is broken down into month by month operations and the achievements are compared with estimates. Where very serious adverse variances occur, management must seek the reason and take immediate corrective action. In this way, the budget prevents irrational spending by over-zealous managers.

For accurate budgeting, it is necessary to know the quantities and prices of inputs and outputs. Thus, by calculating the various costs and examining the revenues in existing rubber firms, it will be easy to guess the future needs of similar firms in future. A careful examination of the cost structure will make it possible to suggest ways to minimize costs and thus maximize profits.

All the firms visited were encouraged to provide information on their major cost items, their total input and output as well as their

- 62 -

selling prices for the various grades of processed rubber. Suffice it to say however that some of the data obtained were subject to certain errors since some of the information obtained from the questionnaires might not be correct. Care must therefore be taken in interpreting the analysis that follows.

Methodology

As mentioned earlier, individual interviews were held with each of the processing firms. The questionnaires prepared for this purpose were used. The products of the industry are rubber sheets, crepe rubber and crumbs. The output figures for the 1974 and 1975 production periods were obtained for each firm. Also recorded were total quantity of products shipped and the volume of sales for the specified period.

The factory size for each firm ranged between 5 to 25 hectares. Land was obtained either by outright purchase, lease or donation. In case of lease, the annual rents paid on the land were obtained.

expenses on labour can be separated into salaries and wages. Salaries refer to payment to the senior staff including items such as Directors fees, while wages are payments made to other categories of workers.

The maintenance of all types of fixed assets is of paramount

importance to the smooth running of each processing firm. The costs of maintaining these fixed assets were therefore included in the analysis. Cost of raw materials which are principally latex and lumps were also included. The costs of other materials for those firms producing sheets and crumbs were also obtained. In most cases, the value of the raw materials employed by mills which obtained their raw materials from the other rubber estates were computed by multiplying the quantity of raw materials by the prevailing prices.

Gross margin for each firm was obtained by subtracting the total operating cost from the gross receipts. The net returns in turn was obtained by finding the difference between the gross margin and total depreciation charged on the fixed assets including interest charges on borrowed funds.

Cost of rubber processing

The cost elements involved in rubber processing can be divided into variable costs and fixed costs. Variable costs are defined as those cost items which change with changes in the level of output. The fixed costs are those costs which do not vary with increase or decrease in the level of output. The cost outlay got from individual firms were computed from the records provided and the results obtained are presented in tables 4.1 and 4.2. In this analysis variable costs include items such as cost of raw materials, water, power and energy, cost of labour and transportation expenses. Other variable cost items are selling and shipping expenses which also include excise duty, sales tax and the harbour dues. Lastly also included are cost of repairs and maintenance, as well as miscellaneous expenses. The latterconsists of administrative and welfare charges.

Fixed costs include items such as Depreciation charges on fixed assets as well as interests on borrowed fund. In most cases, the depreciation on fixed assets had been calculated and presented in form of annual records of each firm, but in a few cases, the depreciation had to be estimated by the author. Where the latter situation prevailed, the straight line method was used to depreciate all the fixed assets in the mills. This method is relatively simple to compute and perhaps this explains its popularity among many workers. The two sets of depreciation currently employed by auditors auditing the accounts of the processors are given below.

The charges on borrowed capital depend on the sources since the interest charges vary from institution to institution.

In 1974, the total cost ranged between ¥149,168 and ¥2,326,410 with an average of ¥868,838 and in 1975, the total cost ranged between

- 65 -

- 66 -

ITEM	1	Ā		В
	Rate (%)	Life span (years)	Rate (%)	Life span (years)
Residential and factory Building	5	20	4	25
Plant and Machinery	10	10	122	8
Bridges and Fences	2	50	11/2	66
Motor vehicles	25	4	33 ¹ / ₂	3
Furniture and Fittings	10	10	1212	8

Source : Firms records during field survey.

₩155,552 and ₩1,696,776 with an average of ₩628,079. Tables 4.3 and 4.4 show the comparative percentages of the various cost items in 1974 and 1975.

Ranging from 33.3% to 83.0%, the raw materials cost constitutes on the average the highest percentages of the operational costs in both years. Other major items of cost are labour cost, selling expenses and fixed costs. These major cost items are discussed in details below:

Cost of raw material

The raw materials for the two years under study can be seen in tables 4.1 and 4.2. It ranged between \$85,792 to \$1,350,525 in 1974 and between \$56,135 to \$912,707 in 1975. The percentages of costs of raw materials are presented in tables 4.3 and 4.4. The high cost of raw materials can be attributed to the fact that the rubber processing industry depends solely on latex as principal raw materials. The cost of latex has been rising for the past five years. In addition, other raw materials such as formic and acetic acid are also expensive. The basic characteristic of the industry is that unlike other industries, rubber processing has no by - products since all natural rubber is turned to intermediate products such as sheets, crepe and crumb which are further manufacture into final goods.

Labour cost

The rubber processing industry can be described as being relatively capital intensive. The industry like most modern industries depends on modern technology. A lot of automation has been acquired in the processing firms. All the units are equipped with different types of processing machines and implements. Although, the processing machines are power driven, the mills engage labour to operate these machines. All categories of labour, unskilled, semi-skilled and skilled were employed.

- 67 -

The intensity of labour use in the industry was measured by comparing the cost of labour (in terms of salaries and wages) with the total cost. The results for the two years 1974 and 1975 are presented in tables 4.5 and 4.6. The percentages of labour cost with respect to total cost ranged between 3.21% to 15.71% with an average of 9.07% in 1974, and in 1975 it ranged from 5.67% to 31.67% with an average of 13.89%. The cost of labour in 1975 could be seen to have almost doubled the cost of labour in 1974. The phenomenal rise in labour costs is attributed to the Udoji salary award which was paid towards the end of 1974. This award substantially increased labour costs and ultimately the total cost of processing rubber. In addition, the award led to scarcity of labour in the agro-industrial sector since most unskilled labourers got better paying jobs in the other sectors of the economy, especially the constructions industry.

It is also remarkable that labour costs were generally the highest in government owned processing units. Based on the employment data in these establishments labour tends to be under utilized. Selling and shipping expenses.

This is another major cost item identified by all processors as constituting a drain on their returns. It ranks relatively high in some of the processing firms. The selling and shipping expenses for all the processing firms for 1974 and 1975 are presented in tables 4.9 and 4.10. It ranged from 0.25 percent to 24.55 percent with an average of 8.81 percent in 1974 and between 0.21 percent to 18.92 percent with an average of 5.78 percent in 1975.

Selling expenses varied from firm to firm. The high cost of selling was mainly due to the shipping operation characteristic of rubber exports. In fact, almost all the processing firms have direct contact with overseas buying agents - to whom these products are shipped and sold. The selling expenses include such items as sales tax, excise duty, harbour dues and marine insurance. The decrease in the percentages of the selling and shipping expenses between 1974 and 1975 is due to the abolition of sales tax on rubber. This cost item is very low in some firms who engaged in little or no shipping abroad since the bulk of their products are sold within the country.

Fixed costs

These cost items include depreciation on all fixed assets such as residential and factory building, motor vehicles, plants and machineries, furniture and fittings. Fixed cost also include charges on borrowed funds.

The high cost of these fixed items can be attributed to the high interests charged on borrowed capital by the commercial institutions as well as high rates of depreciation allowed to processing firms under accelerated depreciation allowances of government. Tables 4.11 and 4.12 show the fixed cost items and their percentages in 1974 and 1975 with respect to the total cost outlay. The depreciation charges ranged between 0.35 percent to 15.41 percent with an average of 5.23 percent in 1974 and from 0.58 percent to 14.75 percent with an average of 5.80 percent in 1975. While charges on borrowed capital ranged between 0.09 percent to 8.04 percent with an average of 1.49 in 1974 and from 0.12 percent to 6.95 percent with an average of 1.96 percent in 1975.

The borrowing of capital is common among the indigenous entrepreneurs within the industry. Also, many of these entrepreneurs got their capital from financial institutions to buy over the processing firms following the indigenization decree.

B. <u>Returns Analysis</u>

The returns to the rubber processing industry are obtained by finding the product of the output and the unit prices of these products. There is no single product price which can effectively represent the price of the various grades of sheets, crepe and crumbs. Besides, prices of rubber fluctuate from week to week depending on supply and demand conditions in international market.

The Gross receipts which are also referred to as volume of sales

Table 4.1 : Cost outlay for the rubber processing firms in 1974 (

			Variable (lost			-	Z,	Fixed (Cost		-
Firm Code	Raw Material (注)	Fuel, Gas Water and Electricity (H)	Saltries And Wages (注)	Building and Machinery Maintenance (H)	Transport Expenses (H)	Selling and Shipping Expenses (所)	! Miscella- neous Expenses * (≌)	Depr. on Building (単)	Depr. on all Plant and Ma- chimery	Interest on Borrowed Capital (所)	Other** Fixed Cost (开)	Total Cost
L01 L02 L02 L03 L04 L05 L06 L07 L08 L09 L10 L11 L12 L13 L14 L15 L16	1,098,664 846,244 85,792 1,284,268 385,345 1,350,525 874,263 1,096,679 295,344 177,928 176,292 194,728 252,488 252,488 254,944 112,767 464,198	27,560 11,668 14,368 16,319 11,940 125,579 29,939 66,433 40,149 20,007 12,876 3,906 2,235 34,297 1,450 55,085	87,208 85,588 24,968 53,325 137,306 189,505 76,416 109,712 60,122 61,667 64,785 40,017 18,841 50,042 15,553 106,860	7,077 11,705 5,117 19,560 15,889 132,138 58,919 25,479 10,522 14,065 9,687 27,966 1,942 13,652 12,405 39,362	18,436 23,542 7,052 6,178 3,402 75,513 1,117 15,020 4,848 9,916 6,853 24,005 908 - 2,944 8,022	51,979 76,814 9,835 260,909 2,519 299,215 70,362 32,437 136,240 119,361 63,119 72,713 25,530 36,200 542 1,839	117,500 105,867 7,074 13,077 49,478 37,671 33,863 26,099 49,685 8,267 20,253 197,963 13,190 37,106 979 10,638	6,435 7,427 2,212 3,752 3,814 23,259 6,572 26,781 7,886 11,658 22,300 8,086 5,985 49,389 815 51,314	4,195 41,830 1,597 351 4,200 47,052 17,011 27,996 17,198 36,062 9,682 8,809 - 1,223 -	68,682 17,771 14,209 6,601 30,740 1,000 87,497 658 - - - 3,670 -	12,526 10,486 4,463 1,764 1,423 17,213 3,801 4,727 12,333 27,169 26,603 6,942 - - 490 -	1,480,262 1,238,938 176,687 1,659,503 521,917 2,328,410 1,173,263 1,518,860 634,985 486,100 412,450 585,135 322,789 475,630 149,168 737,318
Average	559,404.30	29,488.18	67,619.68	25,342.81	12,984.68	77,475.8	45,544.37	14,855.3	1 13,575.3	8 14,426. I	75 8,121	.25 868,838.44

* Miscellaneous expenses are detailed out in Tables 4.3a and 4.4a.

or Other fixed cost includes Depreciation on Jetty, Vehicles, Fences, Bridges, Furniture and Fittings.

counter: Field survey (November - December) 1976.

Lable 4.2 : Cost Untity to the Rubbar processing firms in 1995 (H)

- 72 -

	Variable Cost								Fixed Cost			
Firm Cole	Raw Maternal (≆)	Fuel, Gas Water and Electri- city (H)	Salaries and Wages (N)	Building End Machine ry Mainte- nance (N)	Transport Expenses (N)	Selling & Shipp i ng Expenses (H)	liscella- neous (₩)	Depr. on Building (#)	Depr. on All Plant and Machinery (₩)	Interest on Borrowed Capital (時)	Other** Fixed Cost	Total Cost (音)
A01 A02 A03 A04 A05 A06 A07 A08 A09 A10 A11 A12 A13 A14 A15 A16	701,811 624,264 83,865 912,707 176,761 893,601 558,440 206,543 362,208 56,135 208,482 NA 96,800 192,488 91,670 289,662	22,960 15,422 16,504 21,819 9,721 115,774 15,566 23,190 47,438 19,060 14,908 NA 224 29,294 2,615 72,484	95,900 83,343 30,316 61,850 34,019 231,593 80,497 65,392 72,827 62,645 69,256 NA 26,481 71,529 9,714 134,000	18,504 21,498 2,493 8,156 7,372 147,228 27,289 15,773 18,182 9,579 NA 1,250 25,172 8,115 73,047	21,774 18,878 5,695 6,706 2,621 71,909 2,450 2,436 5,029 14,998 7,752 NA 975 2,207 6,029	84,228 36,038 7,073 56,536 15,227 57,648 73,837 17,194 138,654 5,058 59,054 NA 4,734 30,755 588 1,313	61,112 129,595 10,722 17,489 47,130 49,418 35,126 20,793 41,021 9,419 42,559 NA 13,033 51,453 1,024 10,166	5,649 7,984 2,101 3,950 3,538 23,259 31,929 28,874 9,040 6,108 8,286 NA 9,036 62,878 1,118 43,912	6,935 35,472 403 390 3,360 53,927 11,469 36,299 29,443 5,885 1,278 NA - - 1,677	59,212 18,934 9,604 	12,427 14,869 4,017 1,960 1,134 22,359 7,056 1,651 10,738 317 4,493 NA - - 435 -	1,090,512 1,010,297 172,793 1,091,563 307,292 1,696,776 873,708 452,849 733,033 197,807 425,647 NA 155,552 463,569 119,171 630,613
Average	363,695.80	28,465.27	75,290.8	26,843,8	11,297.2	38,929.13	36,004.00	16,510.8	12,702.40	12,908.4	5,430.40	628,078.80

* Miscellaneous expenses are detailed out in Table 4.3a and 4.4a. Sources : Field survey (November ; December) 1976

** Other Fixed cost includes Depreciation on Jetty, Vehicles, Fences, Bridges, Furniture and Fittings.

N.A. - not available.

- 73 -Table 4.3 : Cost Outlay for the rubber processing firms in vercentages 1974

	_		Variable	e Cost				F	ized Cost	
Firm Code	Raw Material	Water Power Energy	Labour Cost	Repair and Laintenance	Transport Exponses	Selling and Shipping Expenses	Miscallaneous Expenses	Interest on Loan	Depreciation on Building, Equipment and Other fixed Cost	Total Cost
401 402 403 404 405 406 407 408 409 410 411 412 413 414 415 416	74.22 68.30 48.56 77.39 73.83 58.00 74.51 72.20 46.51 36.60 42.74 33.28 78.22 53.60 75.60 62.96	1.86 0.94 8.13 0.98 2.29 5.39 2.55 4.37 6.32 4.11 3.12 0.67 0.07 7.21 0.97 7.47	5.89 6.91 18.95 3.21 7.15 8.14 6.51 7.22 9.47 12.69 15.72 6.84 5.84 10.52 10.43 14.49	0.48 0.94 2.90 1.18 3.04 5.68 5.02 1.68 1.66 2.89 2.35 4.78 0.60 2.87 8.32 5.34	1.25 1.90 5.57 0.37 0.65 3.24 0.10 0.99 0.76 2.04 1.66 4.10 0.28 - 1.92 1.92 1.09	2.16 6.20 3.97 15.72 0.48 12.85 6.00 2.13 21.45 24.55 15.30 12.43 7.91 7.62 0.36 0.25	7.94 8.54 3.99 0.79 9.48 1.62 2.88 1.72 7.82 1.70 4.91 33.83 4.09 7.80 0.66 1.44	4.64 1.43 8.04 	1.56 4.82 4.68 0.35 1.82 3.76 2.33 3.92 5.89 15.41 14.20 4.07 1.82 10.38 1.69 6.96	100 100 100 100 100 100 100 100 100 100
Average	61,03	3 53	9.07	3 11	1.52	8,81	6.20	1 49	5.23	100

Source : - Computed from Table 4.1

- 74 -

Table 4.3a Miscellaneous cost outlay in 1974 (到)

Firm Code	Auditor's Fees Legal Charges & Professional Charges (鞅)	Cffice Expenses Telephone, Post- ages, Entertain- ment fees and Advert (N)	Insurance and Housing Scheme (執)	Nedical Expenses and NF. (K)
A01 A02	4,274 2,578	5,035 2,486	1,052	6,910 669
A03	1,390	1,219	888	1,941
A.04	1,206	980	2,570	1,900
A05	2,000	561	1,142	33
A06	2,000	965	19,338	15,368
AO7	10,728	4,902	5,263	1,607
A08	2,099	3,800	7,559	1,024
£09	4,436	6,970	20,471	15,835
A10	1,200	1,149	2,553	3,344
&11	1,200	3,555	2,345	13,153
A12	1,200	1,616	5,765	16,020
413	2,000	886	896	1,398
A14	3,000	9,251	14,826	6,839
£15		233	-	746
A16	-4-	1,875	7,049	1,714

Source :- Field survey (November - December) 1976.

To de a se Cost ouday for no resalan fires in percentage (1)

÷

			Variable	Cost %				Fix	ed Cost %	
Pirm Code	Raw Material	Water Power and Energy	Labour Cost	Repair and Maintenance	Transport Expenses	Selling and Shipping Expenses	Miscellaneous Expenses	Interest on Loan	Depreciation on Building Equipment & Other Fixed Cost	Total Cost
.01	64.36	2.11	8.79	1.70	2.00	7.72	5.30	5.43	2.29	100
.02	61.79	1.53	8.25	2.13	1.87	3.57	12.33	1.87	6:17	100
03	48.53	9.55	17.54	1.44	3.30	4.09	6.21	5.56	3.77	100
0.4	63.61	2,00	5.67	0.75	0.61	5 18	1.50		0.58	100
.05	57.52	3.16	11.07	2.40	0.85	1.96	15.34	2.09	2.61	100
06	52.66	6.82	13.65	8.68	4.24	3.16	2.31	2.01	5.87	100
07	63.92	1.78	9.21	3.12	0.28	8.45	4.02	3.44	5.77	100
0E	45.61	5.12	14.44	4.20	0.54	3.80	4.59	6.95	14.75	100
0.0	49.41	6.47	9.94	2.15	0.69	18.92	5.30	0.12	6.70	100
10	28.38	9.64	31.67	9.19	7.58	2.56	4.76	-	6.22	100
11	48.98	3.50	16.27	2.25	1.82	13.87	10.00	-	3.30	100
12	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
13	63.23	0.14	17.03	0.83	0.63	3.04	8.58	1.94	5.8	100
14	41 52	6.32	15,43	5.43		6,63	11.10	-	13.56	100
15	76.92	2.19	8,15	6.81	1.85	0.49	0.86	-	2.73	100
116	45.93	11.49	21.25	11.58	0.96	0.21	1. 171		6.96	100
verage	55.42	4.79	13.89	4.18	1,81	5.78	6.36	1,36	5.80	100

Source : Computed from Table 4.2 N.A. Not available

-

Table 4.4a Miscellaneous cost outlay in 1975 (H)

Firm Code	Auditor's fees Legal Charges and Professional Charges (₩)	Office Expenses Telephone, Post- ages, Entertain- ment fees and Advertisement (新)	Insurance and Housing Scheme (社)	Medical Expenses and N.P.F. (N)
401	3,265	4,112	7,015	3,200
A02	3,400	1,417	1,866	914
103	1,830	989	4,439	2,080
A04	2,172	786	2,154	2,739
405	2,000	627	894	437
106	2,000	1,376	24,790	21,252
<u></u> ≜07	3,323	3,935	7,713	2,245
±08	1,564	2,878	3,400	663
409 604	5,000	6,993	13,514	13,970
410	1,200	1,227	2,176	4,191
4.11	1,200	4,652	2,224	16,588
L12	n.a	n.a	n,a	n.a
L13	2,000	1,292	1,172	1,519
A14	4,000	13,158	25,590	5,674
L15	-	344	-	680
A16	-	714	7,267	2,185

Source : Field survey (November - December) 1976 N.A. Not Available

10.040.040

			4
F ir m Code	(a) Labour Cost (秋)	(b) Total Cost (34)	(a) 35 % of (b)
401	87,208	1,480,262	5.89
102	85,588	1,238,938	6.91
403	24,968	176,685	14.13
A-04	53,325	1,659,503	3.21
L05	37,306	521,917	7,15
406	189,505	2,328,410	8.14
L07	76,416	1,173,263	6.51
408	109,712	1,518,860	7.22
109	60,122	634,985	9.47
610	61,667	486,100	12.69
-11	64,785	412,450	15.71
-12	40,017	585,135	6.84
413	6,841	322,789	5.84
A14	50,042	475,630	10.52
-15	15,553	149,168	10.43
416	106,860	737,318	14.49
Average	67,619.68	868,838.44	13.89

Table 4.5. Labour cost in relation to total cost in 1974

Source: Field survey (November - December) 1976.

7 irm Code	(a) Labour Cost (衽)	(b) Total Cost (辞)	(a) as % of (b)
£01	95,900	1,090,512	8.79
102	83,343	1,010,297	8.25
£03	30,316	172,793	17.54
04.	61,850	1,091,563	5.67
L05	34,019	307,292	11.07
106	231,593	1,673,517	13.84
407	80,497	873,708	9.21
£98	65,392	452,849	14.44
604	72,827	733,033	9.94
<u>10</u>	62,645	197,807	31.67
411	69,256	425,647	16.27
112	n.a.	n.a.	n.a.
L13	26,481	155,552	17.03
<u>4</u> 14	71,529	463,569	15.43
15	9,714	119,171	8.15
416	1:34,000	630,613	21,25
Average	75,290.8	628,078,80	9.07
			and the second

Table 4.6 Labour cost in relation to total cost in 1975

Source Field survey (November - December) 1976.

N.A. Not Available

F ir m Code	(a) Raw material Cost (₩)	(b) Total Cost (a (诗)) as % of (b)
4.01	1,098,664	1,480,262	74.22
L02	846,244	1,238,938	68.30
403	85,792	176,685	48.56
404	1,284,268	1,659,503	77-39
L05	385,345	0 521,917	73.83
406	1,350,525	2,328,410	58.00
407	874,263	1,173,263	74.51
1.08	1,096,679	1,518,860	72.20
604	295,344	634,985	46.51
L10	177,928	486,100	36.60
A11	176,292	412,450	42.74
£12	194,728	585,135	33.28
13	252,488	322,789	78.22
€14	254,944	475,630	53.60
L15	112,767	149,168	75.60
A16	464,198	737,318	62.96
verage	559,404.30	868,838.44	61.03

Table 4.7 : Raw materials cost in relation to total cost in 1974

۲

Source : Field survey (November - December) 1976.

- 79 -

	and the second se	and the second sec	and the second se
F ir m Code	(a) Raw Material Cost (剤)	(b) Total Cost (科)	(a) as % of (b)
£01	701,811	1,090,512	64.36
402	634,264	1,010,297	61.79
£03	63,865	172,793	48.53
1.04	912,707	1,091,563	83.61
A05	176,761	\$07,292	57.52
406	893,601	673,517	53.40
A07	558,440 🗸	873,708	63.92
108	206,543	452,849	45.61
L09	362,208	733,033	49.41
£10	56,125	197,807	28.38
A11	208,482	425,647	48.98
L12	n.a	n.a	n.a
413	96,800	155,552	62.23
414	192,488	463,569	41,52
A15	141,970	169,463	76.92
A16	289,662	6 30, 613	45.93
Average	363,695.80	628,078.80	55.42

Table 4.8: Raw materials cost in relation to total cost in 1975

Source :- Field survey (November - December) 1976

N.A. Not Available

Firm Code	(a) Selling and Shipping Expenses (刊)	(b) Total Cost (辞)	(a) as % of (b)
401	31,979	1,480,262	2.16
A02	76,814	1,238,938	6.20
103	9,835	176,687	5.57
£04	260,909	1,659,503	15.72.
405	2,519	521,917	0.48
406	199,215	2,328,410	12.85
A07	70,362	1,173,263	6.00
80A	32,437	1,518,860	2.13
4.09	136,240	634,985	21.45
A10	119,361	486,100	24.55
ú11	63,119	412,450	15,30
412	72,713	585,135	12.43
413	25,530	322,789	7.91
A14	36,200	475,630	7.62
115	542	149,168	0.36
L 16	1,839	737,318	0.25
Average	77,475.86	868,838.44	8,81

Table 4.9. Selling and shipping expenses in relation to total cost in 1970

Source: Field survey (November - December) 1976.

- 81 -

Firm Code	(a) Selling and Shipping Expenses	(b) Total Cost	(a) as % of (b)
	(新)	(発茶)	S
i-01	84,228	1,090,512	7.72
402	36,038	1,010,297	3.57
A03	7,073	172,793	4.09
L.04	56,536	1,091,563	5.18
A05	15,227	307,292	4.96
4.06	53,648	1,696,776	3.16
407	73,837	873,708	8.45
4.08	17,194	452,849	3.80
409	138,654	733,033	18.92
L10	5,058	197,807	2.52
L11	59,054	425,647	13.87
412	n.a	n.a	n.a
13	4,734	155,552	3.04
414	30,755	463,569	6.63
L15	588	119,171-	0.49
-16	1,313	630,613	0.21
Average	38,929.13	628,078.80	5.78

Table 4.10 : Selling and shipping expenses in relation to total cost in 1975

Source:- Field survey (November - December) 1976.

N.A. Noth vailable

Firm Code	(a) Charges on Borrowed Capital (玲)	(b) Depreciation on Fixed Assets (判)	(c) Total Cost (衽)	(a) as % of (c)	(b) as % of (c)
401	68,682	23,156	480,262	4.64	1.56
A02	17,771	59,743	7,238,938	1.43	.4.82
£03	14,209	15,346 <	176,687	8.04	4.68
A04	-	5,867	1,659,503	-	0.35
A05	6,601	9,437	521,917	1.26	1.82
106	30,740	87,524	2,328,410	1.32	3.76
A07	1,000	27,384	1,173,263	0.09	2.33
80A	87,497	59,504	1,518,860	5.76	3.92
A09	658	37,417	654,985	0.10	5.89
L10	1	74,889	486,100	-	15.41
A1.1	2	58,585	412,450	-	14.20
A12	15-	23,837	585,135	-	4.07
£13	3,670	5,985	322,789	1.17	1.82
114		49,389	475,630	-	10.38
415	2,528	-	149,168	-	1.69
£16	-	51,314	737,318	-	6.96
Average	14,426.75	36,836.06	868,838.44	1.49	5.23

Table 4.11;	Fixed costs	in relation	to the total	cost in 1974
-------------	-------------	-------------	--------------	--------------

Source : Field Survey (November - December) 1976

Firm Code	(a) Charges on porrowed capital	(b) Depreciation on all fixed Assets (N)	(c) Total Cost	a) as % of (c)	(b) as % of (c)
	(#)				
4.01	59,212	25,011	1,090,512	5.43	2.29
A02	18,934	62,325	1,010,297	1.87	6.17
A03	9,604	6,521	172,793	5.56	3.77
404	-	6,300	1,091,563	-	0.58
£05	6,409	8,032	307,292	2.09	2.61
A06	34,062	99,543	1,673,517	2.01	5.87
L07	30,049	50,454	873,708	3.44	5.77
80A	31,495	66,824	733,033	0.12	6.70
A09	862	49,221	452,849	6.95	14.75
A10	-	12,310	197,807	-	6,22
A11	2	14,057	425,647	-	3.30
412	11.48	n.a	n.a	n.a	n.a
A13	3,019	9,036	155,552	1.94	5.8
414		62,578	463,569	***	13.56
415		3,230	119,171		2.73
416	-	43,912	630,613	-	6.96
Average	12,908.4	34,643.6	628,078.80	1,96	5.80

Table 4.12: - Fixed costs in relation to the total cost in 1975

 n_{a} . = The record for 1975 is not available.

Source:- Field survey (November - December) 1976.

1

are therefore calculated by taking a product of prices and different quantities and types of rubber product. In some cases where the volume of sales is not clearly recorded, the gross receipts were computed from either the output or the quantity of rubber shipped plus local sales at the average prevailing price.

The results of the returns analysis are presented in tables 4.13 and 4.14 for the two years under review. The result of the analysis shows that in 1974, twelve firms representing 75 percent of the total number of firms studied recorded positive returns and in 1975, eleven firms representing over 70 percent of all firms studied recorded positive returns. Also in 1974, the positive net returns to the firms ranged between ¥18,595 to ¥828,505 while in 1975, it ranged between ¥25,288 to ¥636,989.

From the analysis, it can be observed that some firms recorded negative net returns. In fact, four out of the total number of firms studied recorded negative net returns in 1974 while the number was also four in 1975. The negative net returns recorded in 1974 ranged between \\$527,449 to \\$46,994 and from \\$483,444 to -\\$74,893 in 1975.

Tables 4.15 and 4.16 present the comparative analysis of gross receipts, total cost and net returns for the industry in 1974 and 1975.

- 85 -

Firm Code	Gross Re ce ipts (判)	Total Cost (袓)	Net Revenue (Ħ)
401	1,540,918	1,480,262	60,655
++ 02	1,301,838	1,238,938	62,900
103	233,157	176,685	56,472
404	1,132,054	1,659,503	-527,449
д05	540,612	521,917	18,695
706	3,156,915	2,328,419	828,505
407	891,112	1,173,263	-282,151
E08	1,810,978	1,518,860	292,118
1.09	981,695	634,985	346,710
£10	439,106	486,100	-46,994
L11	339,850	412,450	-72,600
<u>⊼</u> 12	898,229	585,135	313,094
L15	470,880	322,789	148,091
L14	881,537	475,630	405,907
<i>L</i> 15	258,558	149,168	109,390
A16	820,678	737,318	83,360

Table 4.13 Gross receipts total cost and net revenue in 1974

Source : - Field survey (November - December) 1976.

Firm Code	Gross Receipts (N)	Total Cost (秋)	Net Revenue (Ħ)
1 01	607,068	1,090,512	-483,44
1.02	1,035,585	1,010,297	25,288
1.05	208,117	172,793	35,324
40 ;	1,210,255	1,091,563	118,692
105	338,885	307,292	31,593
1.06	2,335,985	673,517	662,468
1.07	598,596	873,708	-275,112
1.08	339,432	452,849	-113,417
409	850,483	733,033	117,450
<u>-</u> 10	122,914	197,807	- 74,893
2.11	558,653	425,647	133,006
1.12	n,a	n.a	n.a
A13	206,195	155,552	50,643
414 V	679,400	463,569	215,831
-15	151,315	119,171	32,144
s16	663,459	630,613	32,846

Table 4.14 : Gross receipts, total cost and net revenue in 1975

Source : Field survey (November - December) 1976.

N.A. Not Available

Firm Code	Gross Receipts (別)	Variable Cost	Fixed Cost	Total Cost N	Not Returns
401	1,540,918	1,388,424	91,838	,480,262	60,656
402	1,301,838	1,161,424	77,524	1,238,938	62,900
A03	233,157	147,130	29,555	176,685	56,472
A.04	1,132,054	1,653,636	5,867	1,659,503	-527,449
105	540,612	505,879	16,038	521,917	18,695
60A	3,156,915	2,210,146	118,264	2,328,410	828,505
A07	891,112	1,144,875	28,384	1,173,263	-282,151
108	1,810,978	1,371,859	147,001	1,518,860	292,118
209	981,695	596,910	38,075	634,985	346,710
1-10	439,106	411,211	74,889	486,100	- 46,994
411	339,850	333,865	58,585	412,450	- 72,600
±12	898,229	561,298	23,837	585,135	313,094
L13	470,880	313,134	9,555	322,789	145,091
£14	881,537	426,241	49,389	475,630	405,907
A15	258,558	146,640	2,528	149,168	105,590
416	820,678	686,004	51,314	737,318	83,360

Table 4.15 : Gross receipts, cost and net returns for the processing firms in 1974

Source - Field survey (November - December) 1976.

Firm Code	Gross Receipts ℕ	Variable Cost ₩	Fixed Cost ₩	Total Cost ₩	Net Returns 관
401	607,068	1,006,289	84,223	1,090,512	-483,144
A02	1,035,585	929,038	81,259	010,297	25,288
L03	208,117	156,668	16,125	172,793	35,324
404	1,210,255	1,085,263	6,300	1,091,563	118,692
405	338,885	292,851	14,441	307,292	51,593
406	2,335,985	1,563,171	133,605	1,696,776	638,989
407	598,596	792,530	98,299	452,849	-275,112
408	339,432	354,550	80,503	873,033	-117,550
409	850,483	682,950	50,083	733,033	117,450
£10	122,914	185,497	13,310	197,807	74,893
4- 11	558,653	411,590	14,057	425,647	133,006
112	n.a	n "A	n.a	n "a	n.c.
±13	206,195	143,497	12,055	155,552	50,643
414	679,400	400,691	62,878	463,569	215,871
A15	151,315	115,941	3,230	119,171	32,144
Å16 ·	663,459	586,701	43,912	630,613	32,646

Table 4.16: Gross Receipts, Cost and Net Returns for the Processing Firm in 1975

Source :- Field Survey (November - December) 1976.

- 89 -

From these tables, it can be observed that there was a general decline

in net returns to these firms in 1975 in comparison with those of 1974. This decline is attributable to many factors. These factors include nonavailability of raw materials (which in turn reduced the volume of sales). Also, lack of raw materials led some mills to operate below optimum capacity.

Another factor which led to lower net returns in 1975 is the Udoji salary award which in a way reduced the output of the industry because many workers went on strike for months whilst production expenses rose astronomically. This constituted a double drain on the profits of processors.

Finally, table 4.17 shows the net return per tonne of processed rubber in all the processing firms in 1974 and 1975. This can be used in comparing the economic performance of all the firms within the processing industry. It also shows how efficient the firms are. The performances of all the firms in 1974 were higher than the performances in 1975. All the factors that are responsible for this generally poor economic performance by all the firms were mentioned earlier.

- 90 -
| | Net Ret | Net Return Per Tonne | | | | |
|-------|---------|----------------------|--|--|--|--|
| ode | 1974 | 1975 | | | | |
| -01 | 12,52 | -290.70 | | | | |
| -02 | 15.37 | 8.08 | | | | |
| -03 | 81.02 | 61.97 | | | | |
| 104 | -121.48 | 21.31 | | | | |
| L05 | 10,28 | 33.36 | | | | |
| 406 | 106.00 | 95.29 | | | | |
| L07 | 72.29 | -119.72 | | | | |
| 80. | 52.72 | - 92.58 | | | | |
| -09 | 145.68 | 47.94 | | | | |
| 110 🏑 | -41.37 | -187.23 | | | | |
| L11 2 | -51.27 | -137.54 | | | | |
| 112 | 174.62 | n,a | | | | |
| L13 | 185.68 | 76.27 | | | | |
| 414 | 101.43 | 135.23 | | | | |
| u15 | 117.75 | 82.58 | | | | |
| 416 | 33.45 | 14.39 | | | | |

Table 4.17 : Net returns per tonne of processed rubber in all the processing firms in 1974 and 1975

. . .

N.A. Not Available

CHAPTER V

COST FUNCTIONS FOR RUBBER PROCESSING INDUSTRY

A. Preliminary considerations

During the last decade, a number of scholars have carried out some work on cost functions in Agricultural production. One of such studies is by Drs. Oni and Olayemi on the cost and returns in cocoa production in the Western state of Nigeria.⁽⁴⁵⁾ In that study, the authors attempted to derive a functional relationship between cocoa acreage and unit cost of production. They assumed a perfectly competitive resource market and applied the quadratic equation to obtain the minimum point of average cost. This point of minimum average cost was said to represent the long-run equilibrum size of cocoa holding. The authors however, presumed that the result would be of limited empirical value for two reasons: firstly, the derived equilibrum size was based on the assumption of the static concept of perfect competition and secondly, there was a marked heterogeneity in the farm population being studied.

Another relevant study was carried out by Akinwumi in 1970⁽⁵⁾. This study was concerned with the Economics of Maize Production in Oyo division in Western State of Nigeria. The author estimated the cost-output relationships for both early and late maize by fitting five functions namely; Linear, Double-log, Inverse Semi-log, Quadratic and square root. The results obtained showed that there was little or no correlation between output of early maize and total cost because the coefficients of multiple determination (R_s^2) were very low. This low correlation was attributed to the ommission of some important explanatory variables in the model as well as the inherent errors of measurements on the field.

A more recent work was that of Adeniyi, $\binom{3}{3}$ which focussed on the Economics of Irrigated rice production in Bida division, North Western State of Nigeria. This study estimated some cost-output relationships for irrigated rice production in Fadeggi and Edozhigi local government areas of Niger State. The approach and findings of this study were similar to those of Akinwumi since the coefficients of multiple determination (\mathbb{R}_{s}^{2}) were generally low (probably due to ommission of some important factors). All three studies dealt with on-the-farm agricultural production cost

functions in different parts of the country. The major objective of this chapter is to estimate the cost functions for an agro-allied industry - the rubber processing industry.

Certain assumptions have to be made in this analysis. The assumptions are as follows:-

(a) That in all the rubber processing firms studied, all the cost items are similar for the same kinds of operations. Thus the same equipment, raw materials etc. are used to varying levels for producing rubber sheets in each firm which produced sheets etc.

- (b) That based on the above assumption, the cost functions derived should be similar for all firms.
- (c) That cost functions derived should be valid,
- (d) That the period of two years (1974 and 1975) being analysed in this study is too short to permit major changes in the sizes of plants. This means that short-run analysis is more relevant.

B. The Basic model

Cost-output relationship

In the empirical estimation of cost functions for the rubber processing firms, the cost (C) which was specified in this study as (i) Total variable cost and (ii) Total cost was related to processed output (Q). For each relationship i.e. Total variable cost against Total output and Total cost against Total output, the following six functional forms were tried. E.Q. (1) $C = bo + b_1Q$ ------ Linear E.Q. (2) Log $C = bo + b_1 logQ$ ------ Cobb-Douglas E.Q. (3) Log $C = bo + b_1Q$ ------ Exponential

E.Q. (4) $C = bo + b_1 log Q$ ------Semi-log E.Q. (5) $C = bo + b_1 Q + b_2 Q^2$ -----Quadratic E.Q. (6) $C = bo + b_1 Q + b_2 Q^{\frac{1}{2}}$ -----Square root. Where C represents either the Total variable cost or Total cost and Q the total output of processed rubber.

C. Data sources and empirical results

The data were collected from 16 firms for two years period - 1974 and 1975, a period too short for the industry or each firm to change its scale of production. The six functions as stated earlier were fitted separately to each of the two yearly data and the empirical results obtained for total average cost against total output are summarized in tables 5.1 and 5.2 while that of Total cost against total output are presented in tables 5.3 and 5.4.

Considering the results of the six fitted equations for the two separate cost-output relationships, the total variable cost results have been chosen as the cost functions for the rubber processing industry mainly for the following reasons;

- (i) the period of study is regarded as a relatively short period during which the firms could not have made significant changes in their scale of production,
- (ii) the value of the coefficients of multiple determination (R^2)
- (iii) the significance of the regression coefficients as increased
 by F test.
- (iv) the relationship between the regression coefficients and their

standard errors of estimate (a rule of thumb is that the regression coefficient must at least be one and half times greater than their standard errors of estimate), and whether or not the equations contradict a priori expectations with reference to signs of the parameters.

The Cobb-Douglas and semi-log functions are linear and so it is not possible to derive an optimum level of output. Therefore, the quadratic function, which also performed reasonably well has been selected as the lead equation.

The lead equation (Quadratic) for the total variable cost against total output in 1974 is prevented below.

E.Q. (7) = 201.496 + 0.0380Q - 0.0000028Q² (0.02733) (0.0000017) $R^2 = 0.425$ F-Value = 1.480** Note in brackets = Standard error

* = F-Value significance at 10%

For 1975, the lead equation (Quadratic equation) is presented below.

E.Q. (8) C = $386.173 + 0.01105Q - 0.0000009Q^2$ (0.009034) (0.0000005) R² = 0.331, F-value = 0.741

Note in bracket = standard error

Lable 5.1 : 1974 Cost functions for Rubber Processing Industry TVC/C

					and the second se	the second second second second
QUATION	Dependent Variable	Constant Term	Res Co	gression efficient	R ²	F-Ratio
LINEAR	TVC	239 .857	0.009	9570	0.327	1.673
EXPONENTIAL	TVC+	2.365	0.000	02Q 001)	0.365	2.164*
EWI-LCG	TVC	6.947	77.346 (47.315	iq ⁺)8)	0.400	2.672*
CUELE LOG CCEB-DOUGLAS	TVC+	1.947	0.146 (0.078	550 ⁺ 320	0.448	3.507*
Q JADRATIC	TVC	201.496	0.0380Q 0.02733)	-0.0000028Q ² (0.000004)	0.425	1,460*
QUARE ROOT	TVC	86.525	0.04297Q 0.04672)	5.980Q (5.251)	0.433	1.502
the second	A REAL PROPERTY AND ADDRESS OF TAXABLE PARTY.	the second s	and the second sec	the second s	the second se	The second secon

Figures in brackets represent standard errors of the coefficients

Note * = Significant at 10%

+ = log of the variable

	-					
QUATION	Depend- ent Variable	Constant Term	Regressi Coefficie	on ents	R ²	F-Ratio
LIMEAR	TAC	402.620	0.02858Q (0.02292)	Jor I	0.327	1.555
EXPONENTIAL	TVC ⁺	2.585	0.00004Q (0.00002)		0.458	3.447*
E I-LOG	TVC	698,141	(113.636Q ⁺ (113.173)	1	0.264	0.974
CORE-LOG (CORE-DOUGLAS)	TVC	3.023	0.164200 ⁺ (0.10468)		0.399	2.460*
QUADRATIC	TVC	386.173	0.001105Q (0.009034)	-0.0000009Q ² (0.0000005)	0.331	0.741
QUARE ROOTS	TVC	320,801	0.6625Q (0.01204)	+3.83124Q ² (1.200726)	0.338	0.775

Table 5.2 : 1975 Cost functions for rubber processing industry TVC/Q

In bracket = standard error

- + = log of the variable
- * = Significant at 10%

		-				1
EQUATION	Dependent Variable		Regr Coeffic X(Q)	ression Hients	R ²	F-Ratio
LINEAR	TC	267.872	0.0	00000510	0.225	0.745
EXPONENTIAL	TC [*]	2.410	0.0	000028 Q 0000009)	0,286	1.248
EMI-LOG	TC	-88.455	56.0	087Q ⁺ 339)	0.270	1.097
DOUBLE-LOG COBB-DOUGLAS	TC ⁺	1.696	0.1 (0.0	1215Q ⁺ 08237)	0.342	1.854*
QUADRATIC	P.S.			-	-	-
BQUARE ROCT	TC	64.864	-0.00001Q (0.00002)	0.08487Q ¹ (0.012707)	0.286	0.583

Table 5.3. 1974 cost functions for rubber processing industry TC /O

In brackets = Standard error

- + = log of the variable
- * = Significant at 10%



Table 5.4 : 1975 cost functions for rubber processing industry IC/Q

In brackets = Standard Error

*

+ = Log of the variable

= Significant at 10%

It would be seen from table 5.1 that the coefficients of multiple determination (R s) were generally low (ranging from 0.327 in linear to 0.448 in the Cobb-Douglas equation for 1974). Also in table 5.2, the values of R² ranged from 0.264 in semi-log function to 0.458 in Exponential equation. This in essence indicates that for 1974, between 33 percent and 45 percent of the variability in the total variable cost could be explained by output of processed rubber alone whereas in 1975, between 26 percent and 46 percent of the variability in cost was explained by variation in processed rubber output. The low values of coefficients of multiple determination (R s) which in fact are not peculiar to this study could be explained. Other factors such as management capability, institutional arrangements, varying levels of capacity utilization which were not incorporated into this model also affect the total variable cost outlay for the industry.

Optimum point of production

operate at its optimum point with respect to a given plant size. However, as far as the industry is concerned, the optimum point to produce is the point where the total variable cost is minimum. Given the various cost functions, what will be the optimum point of operation for each rubber processing firm? In order to answer this question, the first derivative of the lead equation (Quadratic) was equated to zero and solved for Q. Then the second derivative was examined for its sign, since when it is positive, a minimum point is established but when it is negative, a maximum point is reached on the curve.

The following results obtained for the two years are presented below.

For 1974
TVC = 201.490 + 0.0380Q - 0.0000028Q²

$$(0.02733)$$
 0.0000017)
 $\frac{dTVC}{dQ}$ = 0.0380 - 0.0000056Q
0.0380 - 0.0000056 = 0
0.0000056Q = 0.0380
Q = 0.0380
0.0000056
= 6785.71 tons
Taking the second derivative, we obtain:
 $\frac{d^2 TVC}{dQ^2}$ = - 0.0000056

This is negative and confirms that a maximum has been obtained. Thus in 1974, the optimum output at minimum total variable cost for maximum profit is given above as 6785.71 tons. When compared with the mean output value of the whole industry for the same year (which was 2050 tons). It would be seen that in 1974, the firms in the industry were working for below the optimum capacity. The reason for this hav be due to lack of raw materials as well as lack of sound management practice.

In order to compare the optimal output for the two years under review, the optimum output for 1975 was also estimated

In 1975 TVC = 386.173 \div 0.0115Q - 0.0000009Q² (0.009034) (0.0000005)² dTVC = 0.01105 - 0.0000018Q 0.01105 - 0.0000018Q = 0 0.0000018Q = 0.01105 Q = 0.01105 0.0000018 Q = 6138.89 tons Taking the second derivative, we obtain $\frac{d^2 TVC}{dQ^2}$ = -0.0000018

In 1975, the optimum output of 6138.89 tons was also very much higher than the mean operating output of 2912 tons. The optimal outputs for the two years appear quite close which in essence shows that no changes actually occured in terms of physical investments and processing technologies in plants between 1974 and 1975. The rising variable costs explain in part the drop in optimal output.

In conclusion, the processors in most cases mentioned lack of raw materials and the adverse effects of the Udoji salary award as the main reasons for operating far below the optimum capacity.

WERS

CHAPTER VI

FACTORS LIMITING THE EXPANSION AND ECONOMIC PERFORMANCE OF THE RUBBER PROCESSING INDUSTRY

A. Factors Limiting Expansion

The growth and expansion of the industrial sector of the economy have always been given top priorities by the Federal Government of Nigeria. Towards these ends, the Federal Government has embarked on certain schemes and policies. These policies include: (i) income tax relief (ii) Import duties relief (iii) Custom Duties (dumped and subsidized goods) act (iv) Company tax and (v) Double taxation relief, to mention a few. These policies protected the young indigenous companies from fierce competition coming from well established overseas manufacturers. The rubber industry like many other industries in the country has

experienced considerable fluctuations in size and growth over the years. In the early part of this century, the number of firms in the industry was quite few. Between 1950 - 1965, (a period which could be regarded as the peak growth period), the number of rubber processing firms increased substantially. However, this number dropped in the late sixties following the outbreak of the civil-war. Some firms folded-up due to shortage of raw materials. Even after the civil-war, the situation still remained the same and could even be regarded as worsening. Instead of expanding, the processing industry within the country is in fact now contracting in number of firms. Why is this so?

During the field survey, an effort was made to find out the cause of this apparent decline in the number of firms. The processors were asked to indicate their assessment of the entire industry in Nigeria. They were asked to rank those factors limiting their growth and expansion from the following list of factors.

- (i) Lack of capital
- (ii) Lack of demand for finished products
- (iii) Shortage of raw materials
- (iv) Restriction on importation
- (v) Shortage of technical staff
- (vi) Shortage of management personnel
- (vii) Shortage of equipment, and
- (viii) Shortage of spare parts.

All the sixteen firms studied responded enthusiastically to this section of the questionnaire. Besides ranking the factors, some firms gave additional information which though not listed above could enhance a rapid expansion of the industry. The results of the ranking obtained from all the firms have been summarized in table 6.1. These results were subject to a non-parametric statistical test in order to determine the degree of association among the ranking.

Non-parametric statistics is based on the order of the observations and does not depend on a specific distribution of the data in question. Such a statistics is called a distribution-free statistic. Apart from being a relatively simple statistical technique to compute, it allows the analysis of data which may not be numerically precise but which throw light on the order of importance of the statements being analysed. There are many non-parametric techniques such as the Spearman Rank correlation, the sign test, Kendall coefficient of concordance (W) etc. The most suitable technique for this study is the Kendall coefficient of concordance (W) because it can cope with many rankings without signs.

Apart from being very reliable, this technique gives a measure of the degree of association among a number of observers or judges. Also, in a unique way, it provides a simple measure of agreement among several statements in order of rankings. The mathematical notation used in this analysis is shown below:

 R_1 = Sum of Ranks (statements) $\leq R_1$ = Total sum of Ranks (statements) $\leq R_1$ = Mean of sum of Ranks

$$(R_{i} \frac{\leq R_{i}}{N}) = \text{Deviation from rank mean}$$

$$S = \leq (R_{i} - \frac{\geq R_{i}}{N})^{2} = \text{Sum of squares of deviation from rank mean}$$

$$W = \frac{12 S}{\frac{2 S}{K(N - N)}} = \text{Coefficient of concordance}$$

where

- N = Number of stated statements
- K = Number of Judges (processors)

Using the rankings, the eight statements ranked by the processors are presented in the magnitude of the various sums of ranks (R_1) in table 6.1. The most important statements were given the smallest numerical values. The analysis of ranks were derived and presented in table 6.2

The agreement as to the ranking of the statements among the processors is expressed by W = 0.439. For a meaningful interpretation of the results of this analysis, the test of significance of the calculated value of (W) was carried out by finding χ^2 as given by the formula below $\chi^2 = \frac{S}{1/12^{KN} (N+1)}$

As a rule, when N (the number of statements is larger than 7 as in this study, the above expression is approximately distributed as chi-squared with N - 1 degrees of freedom.

			impo	ortance				
			<u>STATE</u>	ME NTS				8
F ir m Code	I	II	III	IV	V	VI	VE	VIII
A01	1	3	8	7	5	6	2	1
A02	1	8	3	7	5	4	2	6
F.03	8	7	3	5	0,6	1	4	2
A04	3	8	1	17	6	5	4	2
A05	1	3	6	6	7	5	4	2
A06 A07	3 1	8 3	76	5 ©	4 7	6 4	1 5	2
80 A	1	80	3	7	5	6	4	2
CO A	4	3-	6	8	7	5	1	2
A10	1	18	3	7	5	6	4	2
A11	1	7.8	3	7	5	6	4	2
A12	-8-	7	3	4	5	2	1	6
A13 🔨	3	8	7	5	2	6	1	2
A14	4	8	7	5	6	3	2	1
A15	1	8	3	7	5	6	4	2
A16	1	6	3	8	7	5	4	2
	42	104	72	105	87	76	47	43

Table 6.1 : Ranks assigned to the factors affecting the

Source: - Field survey 1976

- 109 -

STATEMENT FIRM	I	II	III	IV	V	VI	VII	VIII
R _i	42	104	72	105	87	76	17	-13
$R_i - \frac{\epsilon_i^R}{N}$	-30.00	32.00	0.00	33.00	15.00	4.00	-25.00	-20.00
$R_i = \frac{\langle R_i \rangle^2}{N}$	900.00	1024.00	0.00	1083.00	225.00	16.00	625,00	131.00
From	table 5.2	, the follo	owing s	statistics	were con	mputed		
$\leq^{\mathrm{R}}_{\mathrm{i}}$	= 576	, c	5					
$\frac{\leq R_{i}}{N}$	= 72	5						
N	= 8							
K S	= ≤(R ₁	- <u></u>)	- 4	1720.00				
Given	the abov	ve informa	tion, w	ve can nov	v compu	te the	coefficio	nt

Table 6.2 : Analysis of ranks assigned to the eight statement by sixteen rubber processors

of concordance using the formula

$$W = \frac{12 S}{K^2 (N^3 - N)}$$

Substituting the calculated statistics, we obtain,

$$W = 56640$$
 = 0.439

which is tested by reference to table A in appendix II of Yates.

If the value of X^2 computed from above expression exceeds the tabulated figure for a particular level of significance and a particular value of df = (N - 1) then the null hypothesis (H_d) that the (K) rankings are unrelated is rejected at that level of significance.

In this study, it was found that W = 0.439 and the significance was determined by applying the expression below:

$$X_{-}^{2} = K(N-1)W$$

= 16(8-1) 0.43

 $\chi^2 = 49.16$

The tabulated X^2 value is 18.48 at 1% level of significance i.e. the calculated X^2 is greater than the tabulated χ^2 showing that there is significant difference.

In concluding, since calculated $\sum_{i=1}^{n} exceeds the tabulated <math>\sum_{i=1}^{n} 2^{i}$ value, the coefficient of concordance is statistically significant at one parcent level, thus indicating a rejection of the null hypothesis that the rubber processors rankings are unrelated to each other. Therefore, there is an agreement in the ranking of the statements by the processors.

Interpretation of the Results

The significance of the calculated (W) leads to the careful observation of the order of the magnitude of the various sums of ranks (R_1) . The results of the chi-square and the significance of the coefficient of concordance (W) indicate that the ordering of the statements by the processors are relatively uniform. The least sum of ranks corresponds with the most important statement among all the statements. The highest sum correspond with the least importance factor affecting the expansion of the industry.

Following the above ranking procedure, the order of importance of the eight statements as made by the processors are as follows

- (i) Lack of capital
- (ii) Shortage of spare parts
- (iii) Shortage of equipment and repairs men
- (iv) Shortage of raw materials
- (v) Shortage of management staff
- (vi) Shortage of technical staff
- (vii) Lack of demand for finished products
- (viii) Restriction on importation.

On going through the above ordering of the statements as ranked by the processors, one could easily see that lack of capital with the least sum of ranks is regarded by most processors as the most limiting factor affecting their expansion. Next to this are shortage of spare parts, repair and maintenance workers and shortage of raw materials. The two least important factors are lack of demand for finished products and restrictions on importation.

It should be emphasized however that the significant value of (W) and the result of chi-square do not necessarily mean that the pooled orderings observed by the processors are correct and exact. Infact, there are other factors not listed above but which have considerable ratings among each processor. Such factors however vary from one processor to the other and it was admitted by many processors that once the first three or four limiting factors have been taken care of, others would automatically be solved.

B. Factors limiting the economic performance of the rubber processin / firms

The above factors found to have affected the expansion of the industry were also limiting the economic performance of the existing firms. Though we accept in principle the ordering of the factors as given by the analysis, the present processing operation is afflicted by a number of problems. These problems invariably limit the profitability of the rubber processing firms. These problems include lack of capital, shortage of spare parts i and maintenance workers, shortage of raw materials and labour.

(i) Lack of capital

The major problem confronting the rubber processing firms is the lack of adequate capital. This problem is rampant in most developing countries. In recent times, Nigeria is gradually overcoming this with the mass production of crude oil. The rubber processing industry is relatively capital intensive. The current shortage of capital as claimed by most processors also lead to scarcity of raw materials since most processors need operating capital to purchase latex or other raw materials. As a whole, rubber industry demands a substantial amount of working capital to meet operating expenses such as purchase of raw materials, labour and other inputs. Also, for a steady supply of rubber lumps, the processors have to make contracts with local rubber lump contractors and pay these contractors in advance, Besides, a lot of fund is spent on wages, plant and machinery repairs and maintenance.

Another important aspect is that the large amount of capital is necessary to purchase new equipment since most of the machines are old and obselete. The processors desire to purchase new sets of machines which can process good quality crumbs.

However, not all the processors are equally affected by the lack of funds. Some processors such as government owned mills, corporation owned mills and some processing firms with government and foreign investors are not in short of fund. The indigenous processors are mostly affected by lack of fund since they find it difficult to loan funds from commercial Banks. The second limiting factor is the shortage of spare parts for processin machines and equipment. This problem is however not peculiar only to the rubber processing industry. It is infact a nation-wide one affecting most industries. All the equipment and machines used in rubber processin; are imported. In some cases, processors purchased old but reconditioned machines. These machines breakdown quite frequently and require a large stock pile of spare parts to keep them in operation. Since these spareparts are also imported, it usually takes months before they are made available to the processors. The delay is often attributed to the port congestion and bureaucracy involved in obtaining the licences from the Federal Ministry of Trade and Industry. In addition, difficulties arise in securing good maintenance technicians as repair workmen.

(iii) Shortage of raw materials

This is also a major problem facing the processing industry. Natural rubber in terms of latex lumps and scraps form the bulk of material input of the processing firms. The continuous shortage of raw materials commenced during the war. This condition has persisted long after the civil war as a result of the following reasons.

(A) Low yield of rubber trees

The old age of the existing rubber plantations and those of smallholdings has been responsible for the decline in latex yield. This age factor coupled with unselected species cultivated by most smallholders, the prevalence of root diseases culmilate in poor yield for the rubber trees. Many of the smallholdings were cultivated in the early part of this century and thus overdue for rehabilitation.

(B) Lack of labour for tapping

Another major attribute for the shortage of raw materials is the lack of labour to tap the rubber trees. Since the out-break of the civil war in 1967, when the tappers who were mostly Ibos had to leave, the industry has been faced with non-availability of good rubber tappers. According to a report by a working party on rubber production in Bendel state as a result of shortage of raw materials, five crepe factories out of the existing eighteen mills which were located in the state had to close down between 1967 and 1968.

Also, the returns to the smallholders have been too low when compared with returns from alternate employment opportunities. For example, the tappers prefer to work as garderners in the "oil boom area" rather than stay as smallholder rubber farmers. Thus, during and after the civil war, the job of tapping has been virtually left for women and boys whilst the itinerant tappers have been employed elsewhere.

It should however be noted that the raw material shortage situation is lighter where shortage of tappers is not acute since the tappers are rated like their counterparts in other government owned establishments. Inspite of this attractive renumeration, some rubber plantations are still short of tappers and hence the mills have to work dow capacity.

With regards to non-plantation mills which depend solely on the of smallholders, the lack of latex supply has led to the folding up_some of the establishments.

(iv) Labour problem

Another problem which is though not regarded as a major problem is labour. This is perhaps the industry is relatively less labour intensive. Instead of labour shortage, what is common within the industry among the unskilled labour force is frequent absenteeism and short stay periods by workers. This is attributed to better opportunities outside the rubber processing industry. Due to the prevailing adverse condition in terms of profits to the processors, many could not pay wages that would make them stay in the industry. There is a general shortage of technical and management staff. As a result of the indigenization decree the top posts in the industry are manned by Nigerians. As presented earlier in table 3.6 of chapter three, it could be seen that there were only nine Expatriate in the industry in 1974 and by 1975, this was reduced to eight. They held top management posts in the industry. The shortage of unskilled labour is being experienced on rubber plantations, estates and small holdings.

(v) Marketing problem

Lastly, the processors however stated there is always demand for their finished products which further confirms the mutual exclusiveness of synthetic rubber and natural rubber. But the frequent fluctuations in and low prices of natural rubber make the processors to be cautious in making contract for their products. Infact, some processors have direct contact with the overseas buyers while many sell through local marketing agents.

CHAPTER VII

AN ANALYSIS OF FINANCIAL STATEMENTS AND ESTIMATION OF PROFIT FUNCTIONS OF THE RUBBER PROCESSING INDUSTRY

A. Preliminary Consideration

This chapter examines the financial statements and estimates some profit functions for the rubber processing firms. In the first part of the chapter, some financial ratios are computed for those firms which submitted their Balance Sheets and Income statements for the period of 1974 and 1975. In the second part, some empirical estimates of the profit functions are made. In deriving the profit functions least square multiple regression techniques were adopted.

B. Financial ratios and their use in business management

The term "Financial Ratios" is used to describe significant relationships which exist between two pieces of financial data. These ratios serve many purposes: they can be used to evaluate the financial condition and performance of a business organization. They can assist management in its basic functions such as forecasting and planning. In indicating the economic performance and level of efficiency of the business concerned, they can be used to improve the efficiency or raise the level of profits of the business organization. A number of reasons account for the wide use of ratio analysis. Firstly, the ratios are easily calculated given all the accounting records. The second reason is that these ratios allow easy comparison between similar firms within an industry. Such a comparison gives insight into the relative financial conditions and performances of the firms. They are also readily understood by various interested parties within and outside a firm. In many cases, not all members of the management team are financially oriented and ratios can provide a basic overview for heads of departments, the general manager and the Board of Directors.

Finally, the ratios are helpful in showing the true picture of the firms financial position to all interested parties outside of management. These include creditors who are interested primarily in the liquidity of a firm. Their claim on the firm's assets is short term and the ability of a firm to pay their claim is best judged by means of a thorough analysis of its liquidity. Also the shareholders whose claim may be long-term are interested in the cash-flow and the firm's ability to service debt over the long run. The shareholder may evaluate this ability by analysing the capital structure of the firm, the major sources and uses of funds, its profitability over-time and projections of future profitability.

Other interested parties include bankers, suppliers of basic inputs

and the government. The firm may also use ratios as part of an analysis to determine its own credit worthiness. Above all, in order to bargain more effectively for outside funds, the management of a firm should be interested in all aspects of financial analysis that outside suppliers of capital use in evaluating the firm. An adverse trend in the ratio calls

for immediate corrective action.

Types of financial ratios

A group of financial ratios designed over the years to measure economic performance and efficiency in certain areas of a business organization may be presented under the following headings: Profitability. Liquidity and Solvency. Profitability and solvency are ratios computed from income statement and sometimes from both the income statement and the balance sheet. Liquidity ratios are mainly computed from balance sheet. It should be noted that no one ratio can give sufficient information by which to judge the financial condition and performance of the firm. Only a combination of these ratios can lead to reasonable judgements. In each of these categories, we have several ratios but for this study only those ratios which can be derived from the available data will be discussed.

I. Profitability Ratios

These are a fundamental measure of how efficiently a business

organization is being run and managed. Profitability ratios are of two types: those showing profitability in relation to sales and those showing profitability in relation to investment. Efficiency within the rubber processing industry is measured in this study by

- (a) The rate of return on operating capital; the higher this ratio is, the better.
- (b) ratio of fixed cost to total revenue: this is expected to be small.
- (c) rate of return on volume of sales: there are acceptable, reasonable levels.

(a) Rate of return on operating capital

This is defined as net return divided by total operating cost for the period under review. The rates of return on operating capital are derived for the various firms and the results presented in table 7.1. From the table, it can be observed that in 1974, four firms had over 50 percent rates of return on operating capital while four firms also recorded negative rates of return on operating capital. In 1975, no single firm recorded a rate of return that was higher than 50 per cent whilst four firms recorded negative rates of return. The rates of return ranged from aff; 90 per cent to 95.22 percent

Firm Code	Rate of Retur	turn (in %)	
	1974	1975	
A010	4.37	-48.04	
F102	5.42	2.72	
A03	38.38	22.55	
A04	-31.90	10.94	
A05	3.69 🤇	10.18	
A06	37.49	40.87	
A07	-24.64	-34.71	
A08	21.29	-31.99	
A09	58.03	17.20	
A10	-11.43	-40.37	
A11	-20.52	32.23	
A12	55.78	d.n.a.	
A13	47.29	35.29	
A14	95.22	5.39	
A15	74.44	27.72	
A16	12.15	5.60	

Table 7.1 : Rates of return on operating capital 1974 - 75

ource:- Calculated from firms' records.

with an average of 22.82 percent for 1974 and for 1975 they ranged from -48.04 percent with an average of 3.75 per cent. The sudden post Udoji increases in labour costs and the attendant disruptions in operations account for the generally poorer performances in 1275.

(b) Ratio of fixed cost to total revenue

This ratio gives an indication of how efficiently the fixed assets are being used. The ratio is defined as total fixed cost divided by total revenue for the specified accounting period. The ratios are also expressed in percentages and the results derived for the various firms are presented in Table 7.2. The table shows 1974 that in/the estimated ratios ranged from 0.52 per cent to 17.24 whilst in 1975, the range was from 0.51 percent to 20.97 percent. These derived efficiency ratio serve two purposes: firstly, in both years it could be seen that some firms were more efficient than others. The most efficient firms had smaller ratios while the least efficient ones had larger ratios. Secondly, when we compare how efficient the industry was in 1974 with that of 1975 , it is seen that the industry was less efficient in 1975 for the same reasons presented under profitability analysis.

- 121 -

Firm Code	Katios of fixed co	st to total reve
	1974	1975
01	5.96	13.87
02	5.95	7.85
03	12.60	7.75
04	0.52	0.51
05	2.97	4.26
06	3.75	5.72
07	3,19	13.45
08	8.12	28.97
09	3,88	5.89
10	17.05	10.00
11	17.24	2.52
1.2	2.65	n.a
13	2.05	5.85
.14	5.60	9.25
15	0.98	2.13
16	6.25	6.62
1		

Table 7.2 : Ratios of fixed cost to total revenue

- 125 -

(c) Rate of return on volume of sales

The ratio of net earnings to volume of sales is another measure of profitability. This ratio is usually defined as net earnings divided by value of sales within an accounting period. It demonstrates the effectiveness with which working capital was employed. Some business are slow selling but have high profit margins while others compensate for small per unit margins by the large volume of sales. In each case, the aim is to attain a reasonable level of profit. The calculated ratios presented in table 7.3 indicated that returns on volume of sales were generally low for the two years. The divergence between net earnings and value of sales is a reflection of high operating expenses incurred by the various firms in the industry. Secondly, the low level of capacity utilization raises unit costs of processing and thus reduced the profit margin.

II. LIQUIDITY RATIOS

Liquidity ratios are used to judge a firm's ability to meet shortterm obligations. From then, much insight can be obtained into the immediate solvency of the firm and its ability to remain solvent in the event of adversities. One of the most universal and most frequently used of these ratios is the current ratio.
 1	2	7	-

	1974	1975
	4.41°2	10/0
A O I	4.0	70 6
A 02	4.2	2 1
A02	24.0	12 0
A04	45.5	9.8
A 05	3.5	9.3
A06	26.2	27.4
A07	31.7	46.0
A 08	16.1	33.4
809 A	35.3	13.8
A10	10.7	60.9
A11	21.4	23.8
A12	34,9	31.9
A13	31.4	24.6
414	46.0	31.8
A15	42.3	21.2
A16	10.2	4.9
	0-	

Table 7.3. Rates of returns on volume of sales

Current ratio = <u>Current Assets</u> Current Liabilities

A somewhat more accurate guide to liquidity is the quick ratio or acidtest ratio. This is defined as

acid-test-ratio = <u>current assets - Inventories</u> current liabilities

This matio is the same as the current ratio except that it excludes inventories - presumably the least liquid portion of current assets from the numerator. The ratio in fact concentrates on cash, marketable securities, and receivables in relation to current obligations and thus provides a more penetrating measure of liquidity than does the current ratio.

Due to non-availability of data, the acid test ratio could not be calculated for this study. However, for those firms which provided the necessary data, the calculated current ratios are presented in table 7.4. The higher the ratio, the more is the chance that a firm will be able to meet its immediate obligations and stay bouyant. Firms with a ratio of 1 1 or less cannot be regarded as having a comfortable liquidity status. A close look at this table reveals that for the two years under review, only firms representing 12.50 percent of all the firms studied could be regarded as being in a good financial condition. This is a reflection of rather poor financial management on the part of the firms. In essence, it implies that most of the firms will have to borrow in order to meet their

	Timulali	u Pottos
Firm Code	1974	1975
A02	0.29:1	0.37:1
A03	0.991	0.28:1
A04	0.98.1	0.51:1
A05	- 🔗	0.29:1
A06	1.051	1.02:1
A07	20	0.21:1
A08	5.7 1	4.66:1
A09	0.71:1	0.45:1
A11	-	0.86:1
A14	4.39 1	4.75:1
21		

Table 7.4. Liquidity ratios of some firms in the rubber processing industry

Source : Calculated from the firm's financial statements

- 129 -

most immediate financial obligations.

III. SOLVENCY RATIOS

A business organization is said to be solvent if it can meet its immediate and long-term financial obligation and commitments. Infact, the solvency ratios reflect the position of a business capital requirements which is being supplied by the owners. These ratios are very useful and employed by people both within and outside the business concern. From lender's point of view, solvency measures indicate the kinds of problems the lending body would encounter in recovering their money in the event of business failure. The ratios can therefore have considerable impact on the availability of outside capital to the business. The solvency of a business organization can be estimated using the following formulae:

(a)	Solvency		Net worth
2			Total net assets
(b)	Solvency	-	long term debt
			Capitalization

The net worth/total net assets ratio gives an indication as to how a business concern can cope with its immediate financial needs, while long term debt/capitalization ratio shows the ability of the firm to meet

- 130 -

its future financial needs. This is called ultimate solvency.

Also, solvency ratios may lead to questions as to who is controlling the firms since as creditors supply more and more capital and theraby assume more risk, they can impose restrictive covenants on managerial independence. On the other hand, solvency ratios may indicate that the firm should consider borrowing more money with a subsequent opportunity of increasing returns on their investment. For this study, due to data limitations, only one of the two formulae was employed in measuring solvency ratios. The computed solvency ratios are presented in table 7.5.

Table 7.5 : in the	Solvency ratios of some e rubber processing indu Solvency = Net Wo	<u>selected firm</u> s <u>stry</u> rth	
A.	Total Net	Assets	
1	Solvency ratio in percentages		
Firm Gode	1974 (Sr)	1975 (Sr)	
A02	80.49		
A05	92.09	-	
A06	88.76	69.84	
A 07	57.72	-	
A 08	65.32	68.45	
A14	79.14	92.00	

Source: Calculated from the firms financial statements

The ratios expressed in percentages are derived for some selected firms that made available their financial statements. The net worth/total net assets in addition to a firm's ability to meet its immediate financial obligations further tells us the amount of capital the owners contributed to the business during the years under review. All the firms show that owner's equity capital accounted for over 50 percent of the capital required by the firms to support their net assets. These ratios thus further confirm the dependence of the various firms on personal sources as their major source of capital to finance the mills. It should not normally be difficult for the firms to obtain additional funds from commercial lending houses. Nowever, the results must be interpreted with caution since the items making up net worth and/or total net assets may be in appropriately valued.

In summary, three points should be noted with respect to this financial analysis: firstly, there is considerable disparity in the rates of profits accruing to the rubber firms. While some firms had negative net returns, others are characterized by substantial positive returns. This variation in the rate of return may well be a good index of efficiency among the firms.

The second point concerns the liquidity positions of the rubber

processing firms. For the two years under review, less than 13 percent of the firms were in favourable liquidity positions, thus majority of the firms would have to borrow to meet their immediate financial obligations. This situation is not always tolerated by most businesses.

The third point to note is that some of the processing firms obtain funds from mainly personal sources. Thus reliance on personal savings rather than credit institutions limit their plant expansion.

NERSIT

C. Analysis of Factors Influencing Profits in the Rubber Processing Industry

The costs and returns in rubber processing have been dealt with in details in chapter three. From that chapter, it was clearly shown that net returns were not only highly variable but they were dependent on several factors. Furthermore, the profits in the industry appeared rather low. For instance, in both 1974 and 1975, about 25 per cent of the firms made negative returns. In both years none of the processors made profits up to one million naira. Besides, large variation occured in the net returns of the firms. This large variation made it quite difficult to reach concrete conclusions about the trend of net returns

In order to meaningfully assess the determinants of net profit, a repression analysis was carried out, Four factors were selected as determinants of net profits. These factors were volume of sales, cost of labour, cost of raw materials, and the amount of selling expenses.

 (i) <u>Volume of sales</u> - This is the total naira value of all transactions resulting in sales during the accounting period.
In this study, it mainly consists of the sales of all types of finished rubber products such as crepe, sheets and crumbs. At times, some firms sold scrap rubber to other processin firms but such sales were not taken into consideration because the rubber had not undergone any processing before they were sold.

(ii) <u>Labour cost</u> - This comprised of salaries and wages paid to all categories of workers. It ranged from Director's fees to unskilled workers' wages. Also included in labour cost are fees paid to owners of mills and wages paid to casual workers who are usually engaged for a short time during peak period. As mentioned earlier, the industry is fairly capital intensive although it still engaged a lot of labour. The percentage of labour cost relative to total cost for the two years in the firms have been presented in Tables 4.3 and 4.4.

(iii) <u>Cost of raw materials</u> - The major raw materials consist of rubber lumps, latex and scraps. Other materials are coagulants mainly used by rubber sheets and crumbs processors. Raw materials from the bulk of the total cost in all the processing units as were shown in Tables 4.3 and 4.4.

(iv) <u>Selling and Shipping Expenses</u> - The finished products are mainly exported to overseas countries. The marketing of these products involve a lot of transactions resulting in considerable expenses. Selling expenses consist of items such as port charges, (Marhour dues), marine insurance, shipping (freight) expenses, custor duty and sales tax: although sales tax was abolished recertly by the Federal Government.

I. The basic model

and

A profit function for the rubber processing, industry can be represented implicitly as follows

 $TT = f(X_1, X_2, X_3, 4, U) -----(EQ 1)$

where II = Net profit in Naira,

 $X_1 = Volume of sales in Naira,$

X = Labour cost in Naira,

🏹 = Raw material cost in Naira,

X = Selling expenses in Naira

The explicit form of equation one can be written in linear for as follows -

Y = bo + biXi -----(EQ 2)

where i = 1k, Y = Dependent variable bo = the intercept (Net profit in Naira)

bi's = Regression coefficients

and Xi's = Explanatory variables

Putting it more clearly, we have

 $Y = bo + b_1 X_1 + b_1 X_2 + b_1 X_1 + b_2 X_1 + b_1 X_2 + b_1 X_1 + b_2 X_2 + b_1 X_1 + b_1 X_2 + b_1 X$

bo = The intercept determines the proper level of

profit throughout the entire equation

and $bi - b_4$ = represent the effect on net profit of a one-unit change of each of the Xs

The explicit forms of the profit functions actually estimated included the linear and power forms and the empirical results are presented in the latter part of this chapter.

Hypotheses

The variables used in the regression analysis and the expected effects of each selected factor on net profits are as follows: (1) Volume of sales - It is hypothesized that volume of

sales and net profit would be directly correlated.

- (ii) Labour cost An inverse relationship is expected to exist between net profit and labour cost.
- (iii) Cost of raw materials An indirect relationship was expected to exist between net profit and cost of raw materials.

 (iv) Selling expenses - It was believed that a negative relationship should exist between the two variables.

II. Estimating Procedure

In deriving the profit functions for the rubber processing industry, two equations namely linear and Cobb-Douglas functional forms were fitted to data obtained for 1974 only. The 1974 data were chosen for the following reasons 1975 was regarded by cost of the processors as the worst year in all aspects. Also, the Udoji salary award had a double ne arive effect on profits by raisin the operating costs and simultaneously curtailing total output and revenue accruing to each firm. In fact, in 1975, there were series of strike actions by factory workers and such strikes invariably led to the closure of some mills for several months. In consequence, there were conspicuous reductions in the volumes of outputs from various rubber processing firms with an ultimate fall in net profit har ins. Also, following the Udoji award, many tappers decided to leave the rubber industry for other sectors of the economy where wages were more attractive. This aggravated the perennial problem of shortage of raw materials for the processing industry.

III. Empirical results

The results of both linear and power functions fitted for 1974 data are presented in Table 7.6. It can be seen from the results that all the coefficients of multiple determination (\mathbb{R}^2_{s}) are generally high. Infact, \mathbb{R}^2 ranged from 0.92 to 0.97. This indicates that in 1974 between 92 percent and 97 percent of the variability in net profits could be accounted for by the selected factors. Also, whilst volume of sales had positive relationship with net returns, other factors had negative relationships since infact the net profit decreases as they increase.

Based on its high explanatory power and the test of significance, the linear equation has been selected as the lead equation for the whole rubber processing industry. This linear equation has been presented in equation 3 below.

EQ. 3 $\Pi^+ = 15135.80 + 0.9488X - 0.978X_2$ $(0.06230) \quad (0.0857)$ $- 2.82054X_3 - 1.06221X_4$ $(0.04897) \quad (0.4007)^4$ $R^2 = 0.97$ $F - value = 80.44^*$ * = Significant at 1% level of probability Figures in brackets = standard error.

- 1:10 -

Table 7.6: 1974 Profit Functions for Rubber Processing Industry

Equation Type	Dependent	Constant	Regression coefficients						
	Variable	term	×1	X ₂	x ₃	x ₄	R ²	F-Value	
Linear	п	15135.00	0.9468 (0.06230)	-0.573 (0.0857)	-2.8205 (0.4897)	-1.06221 (0.4007)	0.9699	80.44	
Power (Cobb - Douglas)	II+	-13,170	16.939 (2.4360)	-10.489 (1.6154)	-4.6375 (1.5211)	-0.2096 (0.598)	0.9168	13.176	

+ log form Figures in brackets are standard error In the linear equation stated above, the test of significance for the regression coefficients showed that all the variables $(X_1, X_2, X_3, \text{ and } X_4)$ were significant at the 1% level of probability. Also, the result of the linear equation in no way contradicted all the stated hypotheses with reference to signs of the parameters.

Because of the possibility of auto-correlated errors in some of the selected variables the zero-order correlation matrix was derived and presented in Table 7.7.

Table 7.7.	Correlation	matrix fo	or the	selected
	factors affect	ting prol	its	

	II	X x	x ₂	x ₃	X ₄
II	1.000				
X ₁	0.64156	1.0000			
X2	0.4728	0.8483	1,0000		
X3	0.1473	0.8183	0.6230	1.0000	
X 4	0.0497	0.4118	0.2702	0,5236	1.0000

It could be seen that there is a close association between volume of sales and cost of raw materials on one hand and volume of sales and labour cost on the other hand. The table thus indicates some presence of auto-correlated errors. This led the author to a rerun of the model after deleting one of the two variables that were closely associated. The statistical trial is described in detail in the next section.

Eliminating autocorrelated errors from the model

Realising that when explanatory variables are highly correlated, the regression coefficients cannot be meaningfully interpreted; an effort was made to eliminate the effects of autocorrelation from the profit functions. This was done through a stepwise regression in which the auto correlated variables were dropped in turn from the basic equation. Thus, the following equations were fitted and examined for their ability to explain the relationship between profit level and the factors believed to affect it.

Eq. 4: II = bo + b_1X_1 Eq. 5: II = bo + $b_1X_1 + b_{4/4}$ Eq. 6: II = bo + $b_2X_2 + b_3X_3$ Eq. 7: II = bo + $b_2X_2 + b_3X_3 + b_4X_4$

where all the variables are as specified earlier.

Both linear and Cobb-Douglas were fitted to each model and the Empirical results are as shown below.

Empirical results of the adjusted model

The relationships between net profits and volume of sales are presented in tables 7.8 and 7.9°. In the linear function, it was observed that the coefficient of multiple determination (\mathbb{R}^2) is relatively high. It is 0.65 and shows that in 1974, 65 percent of the net profits could be explained by the volume of sales alone. Also, the F-value is significant at 10% thus indicating the overall significance of the model.

Table 7. 9' presents the results of the combined effects of volume of sales and selling expenses on net profits. These two are grouped together because they did not show any close association. Using the linear function, the coefficient of multiple determination was reasonably high (0.786). This shows that about 78 percent of the variability in net profits in 1974 could be explained by the volume of sales and selling expenses. Also, the F-value is significant at 10%.

Table 7.10 has been included to show the degree of association between the variables namely net profits, volume of sales and selling expenses. The table shows the absence of auto-correlation in this model.

- 1.13 -

1

EQUATION	Dependent Variable	Constant Term	Regression Coefficient X ₁ (VS)	2 R	F-Value
LINEAR	Π	-153946.325	0.27698X 1 (0.09086)	0.646	9.292*
COBE- DOUGLAS	Ī	-16.297	3.44042X ₁ (1.62077)	0.507	4.506*

- $\overline{II} = f(X_1) \text{ volume of sales.}$ In brackets = standard error

 - = Significant at 10% *

2

		1	1			
QUATION	Dependent Variable	Constant Term	Regression X ₁ (VS)	Coefficient	R ²	F-Value
INEAR	π	-303851.691	0.26237X ₁ (0.07831)	0.29703X ₄ (0.12538)	0.776	5.613*
:OBB-)OUGLAS	Π	-30.28989	(1.42805)	(-2.14779X 4 (0.96320)	0.689	4.972*
	NER	II = f(X ₁ , X In brackets = * = Significa	4) standard erro nt at 10%	or		

Table 7 . 9 Profit functions for rubber processing industry in 1974

	50200104 1801013	
II	X ₁	R-
1		8st
0,44563	1	S
0.48088	0.0787	1
	II 1 0.44563 0.48088	II X ₁ 1 0.44563 1 0.48088 0.0787

Table 7.10: Correlation matrix for some of the selected factors

The results of the combined effects of labour cost and raw material cost on net profits are presented in Table 7,11 From the linear equation, it can be observed that the coefficient of multiple determination (0,28) is rather low. This means that only 28 percent of the variability in net profits in 1974 could be accounted for by the combined effect of labour cost and raw material cost. The F-value was also not significant at 10% of probability.

When selling expenses factor was added as a third variable to labour cost and raw material cost, there was an improvement in the coefficient of multiple determination (0.55). This is to say that the three variables accounted for about 55 percent variability Table 7.11 Frofit functions for rubber processing industry in 1974

EQUATION	Dependent Variable	Constant Term	Regression Coefficients		R ²	F-value
			X ₂ (LC)	X ₃ (RC)		
Linear	π	-15896.445	0.29013 X ₂ (0.51053)	0.19415X ₃ (0.24375)	0.28	0.523
Cobb-Douglas	Π	-8.229257	-0.29257 X ₂ (1.62896	2.42205 X ₃ (1.69673)	0.382	1.03

If $f(X_2X_3)$ In brackets = standard error Table 7.12 . Profit function for rubber processing industry in 1677

EQUATION	Dep <mark>end</mark> ent Variable	Constant Term	Regres	R ²	F-Value		
			X ₂ (LC)	X ₃ (RC)	x ₄ (SE)		
LINESR	П	-68747.541	2.76413X ₂ (1.41105)	-),12655X ₂ (0,28138)	-1.55088X (0.83508)	0.547	1.570
Cobb- Douglas	π	-7.732	-0.51242X ₂ (2.40917)	2.40435X (1.77615)	-0.15404X (1.19601)	0.385	0.636

- 148 -

in the net profits. But the F-value test showed that the estimating equation was still not significant at 10% level. The detailed results are presented in table 7.12.

Also for the combined effect of these three factors on net profits. The correlation coefficient matrix is presented in table 7.13. This table shows an absence of any close association between the variables.

Table	7	.1	Э	:	Correlation	(ma	trix	for	the	selected	factors
					exclud	ling	vol	ume	of	sales	

	II	Q ₂	x ₃	x ₄	
II	1				
×_2	0.1778	1			
Х ₃	0.2355	0.0895	1		
X4	0.0459	0.4114	0.1699	1	

In summary, it has been shown in the latter results that volume of sales and selling expenses are the two major factors affecting profit in the industry. The former empirical results were spurious and invalidated by auto correlation. In general, one percent increase in sales volume is expected to lead to about 3.8 percent increase in net profit while a one-percent increase in selling expenses would lead to a decrease of about 2.14 percent in net profit. These findings have serious implications on the policy prescription for the improvement in the economic performance MWERSTN OF BADA

CHAPTER VIII

SUMMARY AND CONCLUSIONS

Summary of major findings

This study was undertaken to examine the performance of an agro-allied industry in Nigeria. Rubber .. processing industry has been selected because of its high contribution to Nigeria's economy. For instance, before the advint of crude oil, natural rubber was the major source of revenue to the Bend I State Covernment and it ranked fourth as a foreign exchange earner for the country. An economic appraisal of the rubber processing industry is necessary at this time when the industry is facing a continuous yearly decline in output. The problems of declining rubber output are attributable to a number of factors among which are shortage of raw materials and a suspected poor performance of rubber processing firms. In addition an economic appraisal of this industry is likely to reveal the major operational and management problems with respect to the existing processing firms. This is the rationale behind this study.

In chapter one, the economic importance of rubber in the economy was shown. Its contributions to the GNP and its role as means of livelihood to many Nigerians were fully discussed. Many Nigerians are gainfully employed in the rubber processing industry, especially in the Bendel State where over 85% of Nigerian rubber output is produced.

The country's rubber exports had increased over the years and her position in the world market was ranked sixth up till 1963. Thereafter the position has been declining. As one of the major agricultural export crops of Nigeria, rubber accounted for about 8% of the country's total export earnings up to 1961.

Also, in terms of employment generation, about 12 - 15 percent of the active population of Bendel State are engaged in the rubber industry. The industrial survey carried out by the Federal Office of Statistics in 1972 showed that about 8000 people (representing 4.7 percent of the country's total industrial labour force) were employed by rubber processing industry. Other secondary benefits include the provision of various infrastructures in the rural areas particularly in plantations and rubber estates.

The second part of chapter one considered the historical development of rubber processing establishments in Nigeria. The origin of rubber and the various processing stages were fully discussed. Government efforts in encouraging rubber production during the early part of this century were welcomed with a great enthusiasm around Bendel and Delta areas of Nigeria. The first sets of people to establish rubber plantations and subsequently rubber processing establishments in the country were in Bendel State, Among these people are Jo

Thomas and the Millers Brothers. These pioneers were followed by many enterprising Nigerians as well as foreigners and state governments.

The old Western region in order to demonstrate the gains from factory rubber processing, established within the region some co-operative rubber processing mills. Between 1950 and 1967 the rubber attracted a lot of processors. As a result of the outbreak of the civil war during which most people had to leave, many factories had to fold up and up till now some of them have not resumed operations.

A brief review of some provious studies concerning the rubber industry was undertaken in chapter two. It was found that most of these previous research work focussed on development programmes concerning the rubber industry. None actually concentrated on the economic appraisal of rubber processing firms.

In terms of research methodology, the area of study included Bendel, Ogun and Ondo States of Nigeria. These three states were selected primarily for three reasons. Firstly, Bendel State is the major rubber producing state in the country and should have the greatest number of rubber processing firms. Secondly, the three states (i.e. Bendel, Ogun and Ondo including Oyo State and some parts of Lagos State) comprised the former Western Region. In that wise, they would all have many common policies, especially in respect of agriculture. Also due to limitations of funds and time more states could not be covered. The addresses of all the processing firms in the three states were obtained from the Federal Office of Statistics. The data used for the study were collected with the aid of questionnaires distributed to the processing firms, and from Federal Office of Statistics and other government institutions.

In chapter three, some background information of the selected firms were presented. These include the pattern of ownership, the amount of capital investment and sources of these capital. The analysis revealed that some of the firms are owned by private Nigerian entrepreneurs while others are jointly owned by the government and some foreign investors.

Sources of capital include personal savings, loans from commercial banks, government sources and funds from foreign agencies. It was observed that quasi-government credit establishments such as NIDB, NAB and NBCI had not contributed much capital to the development of the rubber processing industry. In terms of manpower, all categories of labour were engaged in the industry. These include skilled labour, unskilled labour, technical staff, managerial staff and operatives. As in some other industries, the rubber processing industry is more capital intensive than labour intensive. A lot of capital is invested in the purchase of the processing machine and equipment. Also, it is noted that the expartriates in the industry are relatively few. The industry is currently experiencing a shortage of both technical and managerial personnel. There is no shortage of unskilled labour, although there are cases of frequent absenteeism among this category of workers.

The raw materials consist of rubber latex, lumps and scraps. Firms that process latex into sheets need a lot of coagulant such as formic acid or acetic acid. Other major inputs include water, fuel and electricity. Considerable amount of water is used in the processing of crepe rubber whilst all the processing machines are power driven.

The chapter also discussed the different technological processes involved in rubber processing. The technology involved in the processing of sheets differs from that of crepe, whilst that of crumb rubber combines both methods in addition to the drying process. Unlike some agricultural export commodities, marketing of processed natural rubber at the time of our field work was not undertaken by the Federal or State Government. The processor marketed their products either through local marketing agents or directly to the international market through their overseas marketing agents.

A detailed analysis of costs and returns in the selected firms within the rubber processing industry was given in chapter four. It was noted that in all the processing firms all cost items were similar for the same kinds of operations. Detailed items of costs were analysed and it was revealed that the most important cost item was raw material. Other significant cost items in decreasing order of magnitude are labour cost, selling expenses and cost of all maintenance.

The gross receipts were represented by the volume of sales. The profit levels varied considerably from plant to plant and were generally low. About 25 percent of the firms recorded negative returns in both years. Also, the net returns in 1975 were lower than those of 1974. This decline ' can be attributed to many factors. These include non-availability of raw materials (which in turn reduced the volume of sales). Also, lack of raw materials led some mills to operate below optimum capacity. Another factor which led to lower net returns in 1975 is the Udoji Jalary Award which in a way reduced

- 156 -

the output of the industry because many workers went on strike for months whilst production expenses rose astronomically thus con-

In chapter five, cost functions for the rubber processing industry were estimated. Six functional forms namely; linear, Cobb-Douglas, Semi-log, Exponential, Quadratic and Square root were fitted to two cost-output relationships. These are total variable cost against Total output and Total cost against Total output. The Total variable cost against total output was more relevant to this study since the period of two years (1974 and 1975) being analysed in this study was too short to permit major changes in the sizes of plants.

Although, they fitted the data better, the Cobb-Douglas and Semi-log functions are linear and so it is not possible to derive an optimum level of output from them. Therefore, the quadratic function which also performed reasonably well was selected as the lead equation and was used in deriving the optimum plant size. It was observed that for the two years, the industry was operating far below optimum level of production.

The analysis of some factors affecting the expansion and

economic performance of the processing industry was carried out in chapter six. Eight major factors were ranked by each processor as affecting the expansion and economic performance of the industry. A non-parametric technique (the Kendall coefficient of concordance) was used in ranking these statements.

The result of the analysis showed lack of capital as the most important factor. Next to this are shortage of spare parts, repair and maintenance, workers and shortage of raw materials. In the last part of the chapter, some explanations were made on these rankings. The lack of capital was attributed to the prevailing shortage of raw materials which made the industry to work below optimum capacity thereby lowering their net returns.

A detailed financial analysis of the rubber processing firms was carried out in chapter seven. Financial ratios and their uses in business and management decisions were considered , Profitability ratios, liquidity ratios and solvency ratios were calculated for the firms which provided the necessary data. Due to the limitations and inadequacy of data, not all the ratios could be estimated.

The major findings from the analysis are:

 (i) there is considerable disparity in the rates of profits accruing to the rubber firms. This variation in the rates of profits may well be a good index of efficiency among the firms.

- (ii) For the two years under review, less than 13 percent of the firms were in favourable liquidity positions, thus a majority of the firms would have to borrow to meet their immediate financial obligations. This situation is not always tolerated by most business.
- (iii) Lastly, several of the processing firms obtained funds mainly from personal sources. This reliance on personal savings rather than credit institutions tended to limit their plant expansion.

The second part of chapter seven was concerned with the empirical estimation of profit functions for the industry. A multiple regression analysis involving four selected factors that influenced the returns was carried out. Two functions, namely linear and Cobb-Douglas were fitted. The linear equation performed very well and it was chosen as the lead equation. Though, the results showed that all the hypotheses postulated were valid, there was correlation between volume of sales and labour cost on one hand and volume of sales and cost of raw materials. Then by deleting some of the variables, the author fitted again the same two functional forms. The results showed that, volume of sales and selling expenses are the major factors affecting profits in the industry.

Limitations of the Study

The major limitation of the study concerns the poor quality of some data collected from various sources. Paucity in data coupled with the attitude of some unco-operative processors called for caution in the interpretation and application of the results.

This study has established the rubber processing industry is in a poor economic state. The poor economic state of the industry would be guaged from the financial state of each processors. All of which stemmed from the lack of raw materials and the subsequent under-utilization of existing capacity. The output of the industry has been drastically reduced while production expenses have been increasing. Given this situation, it is important to find out what should be our policy recommendations in our campaign to improve the economic performance of the rubber processing industry. The following specific recommendations are presented for consideration.

The attitude of the processors could be understood for the present state of affairs with regards to shortage of their basic agricultural raw material concerning the industry.

Also, this is a cross-sectional study of the industry which covered only a two-year period, assumption regarding no changes in the physical investments on the industry could be valid.

Despite all these limitations, the study has managed to appraise the economic position of the rubber processing industry. This appraisal might be useful to individual prospective investors, to the Nigerian rubber institute, the newly created rubber commodity Board, all the Tree Crop Divisions of the States Ministries of Agriculture and Natural Resources.

General Policy Recommendations and Conclusions

The various major findings in this study have great implications for the rubber processing industry. There is a need to find ways of improving the present poor economic state of the industry. In dealing with these problems which are mainly :-

(i) Lack of capital, (ii) shortage of both technical and management personnel.

(iii) Shortage of spare parts and (iv) shortage of raw materials, both the rubber processors and the government have important roles to play.

There is need on the part of the processors to improve their individual economic performance. They need to upgrade their management capability. The lack of highly skilled personnel has been reflected in the poor economic efficiency of most of the mills studied. Such personnel will look into some key factors such as labour, management, all types of purchases made by the firms, selling and all marketing matters and the day to day activities of the firms. Also, in the light of the prevailing situation in the whole rubber industry, rubber processors who hope to make reasonable profits must consider ways to increase their capacity utilisation. They must find ways of getting regular supply of natural rubber. In fact, at this stage they need not attempt to expand their plants. They should repair and maintain the existing facilities and make better use of them .

Since capital is the major problem facing the processing firms, the processors must find ways of improving their poor capital position. This can be done by improved relationship with the lending institutions especially government credit institutions. Also, they can improve the situation through sales of shares to interested people and public institutions. Finally, effortshould be made to effect a continuous

- 162 -
strict financial control with regards to accounting practices of some firms.

Steps to be taken by the government

The newly created rubber commodity Board should set up a rubber processing committee to look into the present state of affairs in the processing industry. This committee should be made up of rubber technologists, rubber marketing experts, production economist and some representatives of rubber processors. Such a committee must include a member of the rubber commodity board and should examine all aspects of rubber processing including lumps production and marketing of the processed rubber.

Government should also look into the possibility of establishing rubber lumps marketing centres. The location of such centres should be around existing processing factories. This must be followed by the standardization of rubber lumps into grades. These steps will eliminate the surpluses made by rubber lump contractors over the rubber smallholders. Since returns to smallholders will be based on the quality of lumps produced, rubber lump producers will aspire to produce better quality and therefore earn renumerative revenue.

Furthermore, such establishment will eliminate the practice where by rubber lumps are heavily adulterated to increase their weight in order to make more money. Subsequently, high quality processed rubber will be produced and the frequent breakdown of processing machines caused by foreign particles in these lumps would be reduced.

Above all, the Rubber Commodity Board which will now control the marketing of rubber in the country should try to erase the poor image already formed by the processors towards the defunct Nigeria Produce Marketing Board by making sure that steady returns get to the natural rubber producers at all levels, especially the processors who have been exposed to the rubber international markets. The Board can through the Federal government explore new markets for the Nigerian rubber in the socialist countries and the third world not only for reasons of exports stability but also for reasons of market expansion. For this purpose, the Board can use bilateral, multilateral and other contractual agreements to penetrate these new markets.

- 162 -

REFERENCES

Report to the Government of Nigeria 1. Ackhaust C.G., Natural rubber Production; Etap Report No 1526.

A study of land tenure and financial opportunits 2. Ademehin, H. O. to rubber production under different systems of management in Western State of Nigeria. Unpublished graduate Diploma presented to University of Ife, March 1972.

Adeniyi J. P., Economics of Irrigated Rice Production in Bida 3 Division North Western State of Nigeria, Unpublished M.Sc. Thesis Dept. of Agric. Economics, University of Ibadan July 1973.

Economic Importance of Nigeria Wood Fased 4. Adeyoju, S. K. Industry; N.J.E.S.S. Vol. 11, No. 3, November 1969.

> Economics of Maize Production in Oyo Division in Western State of Nigeria. Unpublished M.Sc., Thesis, Department of Linic. Economics, University of Ibadan, Decomber 1970.

An Economic Appraisal of the Textile Industry in Lagos and Western State of Nigeria, Unpublished M. Phil Thesis, Dept. of Agric, Cons., University of Ibadan, Ceptember, 1976.

The Improvement of Rubber for the Export Trade , Ministry of Trade and Industry, Benin City, 1966.

Economic Aspects of Peasant Rubber Production in Mid-Western Nigeria, Unpublished Ph.D. dissertation, Dept. of Agric. Economics, Michigan State University, Michigan, 1965.

Amadi Emina, I. 7.

Anschel, K. R., 8.

5.

Akinwumi, J.A.

- Akinyosoye, V.O. 6.

- 9.____
- 10. Asiodu, P.C.
- 11. Barlow, C. and Ching C.S.G.
- 12. Bauer, P. T.,
- Bauer, P. T, and Paish, F. W.
- 14. Derry, S. S.,
- 15. Bradford, L. A., and Johnson, G. L.
- Brookson, C. W., Morris, J. E., Wong Mun Yuu.
 Clayton, Eric S.,

18. Cook A. S.,

Problems and prospects of the Nigerian Rubber Industry. N. J. E. S. S. Vol. 9 No. 2; July 1967.

Industrial Policy and Incentives in Ni eria N. J.E.S.S. Vol. 9, No. 2, 1967

<u>A report of the survey on Malayan group</u> processing centres, 1964 Statistics Division, Economics Report No. 1, Rubber Research Institute of Malaya, Kuala Lumpur, Malaysia, April, 1965.

The Rubber Industry Longmans Green and Co. Ltd., London 1948.

Fluctuations in the Income of Primary Producers Economic Journal Vol. LXII, December 1952, pp. 750 - 780.

<u>Cocoa in Western Niceria 1890 - 1940, A</u> study of an innovation in a Developin Economy: Unpublished Ph.D dissertation, University of Michigan, Michigan, 1967.

Farm Management Analysis, New York, John Wiley and Sons, Inc. 1953, pp. 328 -329. Also quoted in K. R. Anschel, Economic Aspects of Peasant Rubber Production in Mid-Western Nigeria, 1965.

Report on Rubber Research in Federation of Nigeria and in the Cameroons: Kaula Lumpur R. R. I. M. August, 1961.

Agrarian Development in Peasant Economies, The Macmillan Company, New York 1964.

Rubber Production and utilization in Nigeria, Technical Memorandum No. 16 (Lagos Federal Ministry of Commerce and Industry, Dec. 1962).

	-	167 -
19.	C.S.N.R.D., 33	33 Farm Management in Africa, The Principle of Production and Planning, Oxford University Press, 1973.
20.	Douglas P. H., and Cobb C. W.,	A theory of Production, American Economic Review Volume 18 (supplement) 1928.
21.	Essang S. M.,	The Marketing of Palm Oil in a Rural Community Unpublished M.Sc. Thesis, Dept. of Agric Economics, University of Ibadan, Ibadan, 1967.
22.	F.A. O.	Agriculture and Development in Nigeria, 1965 - 1980, Rome 1966.
23.	Glass, M. R. and Crtego, A.J., Jr.,	An Analysis of Financial statements of Virginia's Retail Farm Equipment Businesses Virginia Agricultural Experiment station, Bulletin 541, August, 1962.
24.		Variations in financial statements and analysis affecting profits, Virginda's Retail farm Equip- ment Businesses, wirginia experiments station, Bulletin 168, February 1964.
25.	Gromely P. I.,	Notes on the Economics of the Natural Rubber Industry in Nigeria. Seminar Presentation of N.I.S.E.R., University of Ibadan, May, 1971.
26.	Heady, E. O. and Dillon, J. C.,	Resource Productivity, Returns to Scale and Farm size, Iowa State College Press, Ames 1956. Agricultural Productions functions. Iowa State University press, Ames, 1961
27.	Heady, E. O.,	Economics of Agricultural Production and <u>Resources Use</u> Brentice - Hall Inc., En lewood Cliffs. M. J., 1960.
28.	Helleiner, G. K.,	Peasant Agriculture, Government and Economic Growth in Nigeria, Economic Growth centre, Yale University Richard Myvin Inc., Homewood, Illinois.

- 168

29.	Johnson, G. L.	Factor Markets and Economic Development, in Manhenson W. W. (ed) Economics Development of tropical Agriculture, Gainesville, Florida, University of Florida Press, 1968, pp. 03 - 111.
30.		Strategies and Recommendation for Nigerian Rural Development 1969/85, CSNED Report 33 July, 1969.
31.	Johnston, J.,	<u>Statistical cost analysis</u> , Macgraw - Hill Book Company, New York 1958.
32.	Kmenta, J.	Elements of Econometrics Macmillian company, New York, 1971.
33.	Laurent, C. K.	Investment in Nigerian Tree Crops : Small holder production N.I.S.E.R. University of Ibadan C.S.N.R.D - 18 October, 1968.
34.		Acro-Industries in Nigeria N.I.S.E.R. University of Ibadan, Nigeria C.S.N.R.D 25 Burust, 1980
35.	Leftwich, R. H.,	Price systems and Resources Allocations Hult Rinehart and Winston, New York, 1955.
35.	Lloyd, N.	Report to the Government of Nigeria on the Rubber Industry in the Mid-Western Region: U.N.D.P., T.A. 2517, F.A.O. Rome, 1968.
37.	McHale, T.R.	The competition between synthetic and Natural Rubber. The Malayan Economic Review Vol. VI No. 1 April 1961 pp. 23 - 31.
38.	Cbi, M. C.	The Ikpoba Factory A preliminary case study of the economics of Rubber processing in Mid-Western Provinces of Nigeria, E.D.1 Working Paper No. 7, 1965.
39.	Ckereka, O.	The distribution channels in the marketing of

Rubber in Mid Western State of Nigeria, Unpublished M.Sc. Thesis, Department of Agric. Economics, University of Ibadan, June, 1970.

- 40. Olatunbosun, D. <u>Nigerian Governments Policies affecting</u> <u>Investment in Agriculture</u>, N.I.S.E.R. University of Ibadan C.S.N.R.D. - 16 July, 1966.
- 41. Onabanjo O. O. <u>Economic Appraisal of Rubber Production in</u> <u>selected farms of Ogun state of Nigeria</u>, Final year project, Dept. of Agric. Economics University of Ibadan, May 1976.
- 42. Olayemi, J.K. and S.O. Olayide Major Agricultural Exports. Technical Report ASR. No. 752 Dept. of Agric. Economics, University of Ibadan, March 1975.
- 43. Olayide, S. O. and <u>Trends and prospects of Agricultural Exports</u> Olatunbosun D. N. I.S.E.R. December, 1972.
- 44. Oni, S. A.
- 45. Oni, S. A and Olayemi, J. K.

46. Phillip, C. P.,

47. Phillips, A. O.

An Econometric Analysis of the Provincial

and Aggregate supply responses among Western Nigerian Cocoa farmers. Unpublished Ph.D. Thesis, Department of Agric. Economics, University of Ibadan, March, 1971.

Costs and Returns in Peasant cocoa Production, A case study of Western Nigeria. D.F.ESS Vol. 5 ## 2, 1971..

Evaluation of Agricultural Projects, Bulletin of Rural Economics and Sociology, Vol 5 # 1 October, 1970.

Nigerian Industrial Tax Incentives: Import Duty Relief and Approved User scheme, Nigerian Journal of Economic and Social studies Vol. 9, No. 3, November, 1967.

48.		Nigeria's Experience with Income Tax
		Nigeria Journal of Economic and Social Studies, Vol. 10, No. 1, March 1968.
49.		The significance of Nigeria's Income Tax <u>Relief Incentives</u> Nigerian Journal of Decomic and Social studies, Vol. 11, No. 2, July 1969.
50.	Polhamus, L. G.	Rubber Botany, Production and Utilization, Leonard Hill, (Books) Ltd., London, 1962.
51.	Sawyer, J. S.	The rubber industry in Sierra Leone and its prospects. 1964.
52.	Saylor, R. C.	A study of obstacles to Investment in oil palm and rubber plantation. CSNRE /MISER University of Ibadan, 1968.
53.	Sayre, P. S.,	Obstacles to Nigerian private investment. Nigerian Journal of Economic and social studies Vol. 4, No. 1, March 1962.
54.	Schatil, L.	Industrialization in Nigeria , Oxford University Press , Ibadan 1974.
55.	Siegel, S.	Non-parametric statistics for the Behavioural Sciences McGraw-Hill Book Company Inc. New York, 1906.
56.	Smith, J. F. and Cooper, K.	The Financial Management of Agribusiness firms. Special report 26. Agric. Extension Service of Agric. Economics and U.S., Department of Agriculture, University of Minnesota, September, 1967.
57.	Udo, R. K.	British Policy and the Development of Export crops in Nigeria: Nigerian Journal of Economics and Social Studies, Vol. 9, 1967

- 170 -

58.	Warren, H. V.	Economics and Management in Agriculture Dept. of Agric. Economics, Michigan State University Prentice - Hall Inc., Englewood Cliffs New Jersey, 1962.
59.	Wells, J. C.	Appraising an Agricultural project in Northern Nigeria A problem of investments Evaluation, Nigeria Journal of Economic and Social Studies, Vol. 5, No. 1, March, 1963.
60.	W.N.D.P.	Western Nigeria Development Project Project Analysis No. 2, Rubber Department of Economic Development.
61.	Wessels, H. R.	Principles of Financial Analysis: A study of financial Management. The Macmillion company, New York, 1951.
62.	Yates and Fisher:	Statistical tables for biological, agricultural and medical research, published by Oliver and Boyd Ltd., Edinburgh.
63.	Zwanhuisen, M. Th.,	Agricultural Industries for the processing of raw materials in the Eastern Region. F.A.O. Report, Rome, 1963.
1	5	

- 172 -

APPENDIX I

An Economic Appraisal of the Rubber Processing Industry in Bendel, Ogun and Ondo states of Nigeria

Questionnaire

General

- 1. (a) Name of establishment -----
 - (b) When was it established ? -----
 - (c) Total area (acreage) owned by firm

(e) Name other branches (if any) owned by the firm and

location/-----

Form of Ownership

2. Sole Proprietorship ------Partnership ------Private or Public Limited Company ------Government ------Co-operative ----- Statutory Corporation -----

Location

Investment Pattern



- 175 -

Labour

6.	(a)	Category	Nur	nber	Wages and Salaries			
			1974	1975	1974	1075		
	(i)	Nigerian Professional administrative and			285			
		managerial		J	ρ*			
	(ii)	Non-Nigerian Professional	0	F .				
		administrative and	1					
	(iii)	managerial. Clerical and other						
	(iv)	office workers Technical						
	(v)	Unskilled workers						
	5							

(b) Other benefits to workers

N.P.F.	Contribution
Private	Pension scheme
Medica	al Benefits

- (c) What are the main problems you have with labour in this business Shortage of skilled workers Frequent resignation Strike action Low productivity Others (specify)
- 7. Raw materials
 - (a) State your major raw materials and sources

Type	Source	Quantity Purchased				
A.		1974	1975			
(if)						
(iv)						

- 177 -

(b) Finished products

Туре	Quantity	Quantity Produced					
	1974	1975	1974 19				
(i)			25				
(ii)			×				
(iii)		2					
(iv)		Or					
eting	~						
ting To whom do you (a) <u>Domestic b</u> Governmen	sell your prod	ucts?					
ting To whom do you (a) <u>Domestic by</u> Government Other comp	sell your prod <u>ivers</u> anies	ucts?	107 ang				
ting To whom do you (a) <u>Domestic bu</u> Government Other comp Distributor Appointed	sell your prod nyers anies s Agents	ucts?					
To whom do you (a) <u>Domestic bu</u> Government Other comp Distributor Appointed Others (sp	sell your prod ivers anies s Agents ecify)	ucts?					

Countries

(i)



business.

- (d) Do you consult any firm or business consultants?
- (e) Have you organised any in-service-training for your employees?
- (f) What are the basic goals and objectives of your business?
- (g) Is your business cooperating with any other firms in the industry with regards to
 - (i) Procurement of raw materials
 - (ii) Joint processing
 - (iii) Marketing each others products
 - (iv) Staff exchange
- (h) Are you affected by the indigenization decree?
- (i) What are the main problems facing your business expansion?
- (j) In youropinion, is the rubber industry contracting expanding?
- (k) State briefly what you think the government can do to sustain and effect better expansion of the industry.

APPENDIX II

- 180 -

TABLE I : TABLE OF CRITICAL VALUES OF CHI SQUARE*

df	Probability under H _o that x ² = chi square													
	.99	.98	.05	.90	.20	.70	.50	.30	.20	.10	.05	.02	.01	.001
1 2 3 4 5 6 7 9 9 10 11 12 13 14 15	=00016 .02 .12 .30 .55 .87 1.24 1.65 2.09 2.56 3.05 3.57 4.11 4.56 5.23	.00053 .04 .18 .43 .75 1.13 1.56 2.03 2.53 3.06 3.61 4.18 4.76 5.37 5.98	.0039 .10 .35 .71 1.14 1.64 2.17 2.73 3.32 3.04 4.59 5.23 5.60 5.57 7.26	.016 .21 .58 1.05 1.61 2.20 2.83 3.49 4.17 4.86 5.58 6.30 7.04 7.79 8.55	.064 .45 1.00 1.65 2.34 3.07 3.32 4.59 5.38 5.10 5.38 5.10 5.38 5.10 5.38 5.10 5.38 5.10 5.38 5.10 5.38 5.10 5.38 5.10 7.31	.15 .71 1.42 2.20 3.00 3.83 4.57 5.53 5.53 5.39 7.27 8.15 9.03 9.93 10.82 11.72	.46 1.39 2.37 3.36 4.35 5.35 6.35 7.34 8.34 9.34 10.34 11.34 12.34 13.34 14.34	1.07 2.41 3.56 4.88 6.06 7.23 8.38 9.52 10.66 11.78 12.90 14.01 15.12 16.22 17.32	1.64 3.22 4.62 5.99 7.29 8.56 9.80 11.93 12.24 13.44 14.63 15.81 16.98 18.15 19.31	2.71 4.60 5.25 7.78 9.24 10.64 12.02 13.36 14.68 15.99 17.26 18.55 19.81 2J.06 22.31	3.84 5.99 7.02 9.49 11.07 12.59 14.07 15.51 16.92 18.31 19.68 21.03 22.36 23.68 25.00	5.41 7.82 0.84 11.67 13.39 15.03 16.62 18.17 19.58 21.16 22.62 24.05 25.47 25.67 28.26	6.64 9.21 11.34 13.28 15.09 16.81 18.48 20.09 21.67 23.21 24.72 26.22 27.69 29.14 30.58	10.33 13.32 16.27 18.16 20.52 22.16 24.32 26.12 27.38 29.59 31.26 32.91 34.53 36.12 37.70

Table I is a bridged from Table IV of Fisher and Yates Statistical tables for biological, agricultural, and medical research, published by Gliver and Loyd Ltd., Edinburch, by permission of the authors and publishers.